UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, DC 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported) October 29, 2024

BIRCHTECH CORP.

(Exact name of registrant as specified in its charter)

	Commission file number 000-33067		
	Delaware	87-0398271	
	(State or other jurisdiction of incorporation)	(I.R.S. Employer Identification No.)	
	1810 Jester Drive		
	Corsicana, Texas	75109	
	(Address of principal executive offices)	(Zip Code)	
	Registrant's telephone number, includ	ing area code: (614) 505-6115	
	Not applica (Former name or former address, i		
	ck the appropriate box below if the Form 8-K filing is intended to simultaneously satisfied instruction A.2. below):	fy the filing obligation of the registrant under any of the following provisions (see	
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.	425)	
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a	n-12)	
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange	Act (17 CFR 240.14d-2(b))	
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange A	Act (17 CFR 240.13e-4(c))	
Secu	urities registered pursuant to Section 12(b) of the Act: None.		
	cate by check mark whether the registrant is an emerging growth company as defined i (17 CFR 240.12b-2).	n Rule 405 of the Securities Act (17 CFR 230.405) or Rule 12b-2 of the Exchange	
		Emerging growth company \square	
	n emerging growth company, indicate by check mark if the registrant has elected not to ounting standards provided pursuant to Section 13(a) of the Exchange Act. \square	use the extended transition period for complying with any new or revised financial	

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

Birchtech, Inc. (the "Company") previously reported that effective on October 17, 2024, its corporate name changed from "Midwest Energy Emissions Corp." to "Birchtech Corp." pursuant to a Certificate of Amendment to its Certificate of Incorporation filed with the State of Delaware (the "Corporate Name Change").

On November 11, 2024, the Board of Directors of the Company (the "Board") approved a conforming amendment to the By-laws of the Company to reflect the Corporate Name Change. No other changes were made to the By-laws. A copy of the Second Amended and Restated By-laws reflecting this amendment is filed as Exhibit 3.2 to this Current Report on Form 8-K and is incorporated by reference herein.

Item 8.01 Other Events.

Equity Incentive Plans

The Company previously reported that on July 3, 2023 the Board approved and adopted the Company's Amended and Restated 2014 Equity Incentive Plan (the "2014 Plan") and the Company's Amended and Restated 2017 Equity Incentive Plan (the "2017 Plan") which amended the Company's previously adopted 2014 Equity Incentive Plan and 2017 Equity Incentive Plan. Stockholder approval was obtained on December 5, 2023. On October 29, 2024, the Board approved certain non-material amendments to the 2014 Plan and 2017 Plan which amendments were made in connection with the listing of the Company's shares on the Toronto Stock Exchange ("TSX") and graduation from the TSX Venture Exchange to the TSX. Copies of the 2014 Plan and 2017 Plan, as amended on October 29, 2024, are filed as Exhibits 10.1 and 10.2 to this Current Report on Form 8-K and are incorporated by reference herein.

Majority Voting Policy

On October 30, 2024, the Board adopted a Majority Voting Policy. In an uncontested election of directors, any nominee who receives a greater number of votes "withheld" than votes "for" (i.e., the nominee is not elected by at least a majority (50% +1) of the votes cast with respect to his/her election) is required to tender his/her resignation to the Board promptly following the stockholders meeting. The directors will consider the offer of resignation and, except in exceptional circumstances, will recommend that the Board accept the resignation. The Board will make its decision within 90 days following the stockholders meeting and announce it in a press release, including the reasons for rejecting the resignation, if applicable. The nominee will not participate in any deliberations on the resignation offer. The policy does not apply in circumstances involving contested director elections. A copy of the Majority Voting Policy is filed as Exhibit 10.3 to this Current Report on Form 8-K and is incorporated by reference herein.

Item 9.01 Financial Statements and Exhibits.

Exhibit Number	Description
<u>3.1</u>	Second Amended and Restated By-laws
10.1	Amended and Restated 2014 Equity Incentive Plan, as amended October 29, 2024
10.2	Amended and Restated 2017 Equity Incentive Plan, as amended October 29, 2024
10.3	Majority Voting Policy
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)
	2

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Birchtech Corp.

By: /s/ Richard MacPherson Date: November 15, 2024

Richard MacPherson President and Chief Executive Officer

SECOND AMENDED AND RESTATED BY-LAWS OF BIRCHTECH CORP.

(Adopted November 11, 2024)

ARTICLE I STOCKHOLDERS

Section 1. Place of Meetings. All meetings of the stockholders shall be held at such place either within or without the State of Delaware, as shall be designated from time to time by resolution of the Board and stated in the notice of meeting. The Board may, in its sole discretion, determine that stockholder meetings shall not be held at any place, but may instead be held solely by means of remote communication in accordance with Section 211(a) (2) of the Delaware General Corporation Law. If authorized by the Board in its sole discretion, and subject to such guidelines and procedures as the Board may adopt, stockholders and proxy holders not physically present at a meeting of stockholders may, by means of remote communication (a) participate in a meeting of stockholders; and (b) be deemed present in person and vote at a meeting of stockholders whether such meeting is to be held at a designated place or solely by means of remote communication, provided that (i) the Corporation shall implement reasonable measures to verify that each person deemed present and permitted to vote at the meeting by means of remote communication is a stockholder or proxy holder; (ii) the Corporation shall implement reasonable measures to provide such stockholders and proxy holders a reasonable opportunity to participate in the meeting and to vote on matters submitted to the stockholders, including an opportunity to read or hear the proceedings; and (iii) if any stockholder or proxy holder votes or takes other action at the meeting by means of remote communication, a record of such vote or other action shall be maintained by the Corporation.

Section 2. <u>Annual Meeting</u>. The annual meeting of the stockholders shall be held on such date and at such time and place as the Board of Directors may designate. The date, place and time of the annual meeting shall be stated in the notice of such meeting delivered to or mailed to stockholders. At such annual meeting the stockholders shall elect directors and transact such other business as may properly be brought before the meeting.

Section 3. Special Meetings. The annual meeting of the stockholders entitled to vote may be called by the Chairman, if any, the Chief Executive Officer, the President or any Vice President, the Secretary or by the Board, and shall be called by any of the foregoing at the request in writing of stockholders owning in the aggregate not less than 25% of the entire capital stock of the Corporation issued and outstanding and entitled to vote. Such request shall state the purpose or purposes of the meeting. A special meeting requested by stockholders shall be held at such date, time and place within or without the State of Delaware as may be fixed by the Board; provided, however, that the date of any such special meeting shall not be more than 90 days after the request to call the special meeting is received by the Secretary of the Corporation. Notwithstanding the foregoing, a special meeting requested by stockholders shall not be held if either (a) the Board has called or calls for an annual meeting of stockholders to be held within 90 days after the Secretary of the Corporation receives the request for the special meeting and the Board determines in good faith that the business of such annual meeting includes (among any other matters properly brought before the annual meeting) the business specified in the request or (b) an annual or special meeting that included the business specified in the request (as determined in good faith by the Board) was held not more than 90 days before the request to call the special meeting was received by the Secretary of the Corporation. A stockholder may revoke a request for a special meeting at any time by written revocation delivered to the Secretary of the Corporation, and if, following such revocation, there are unrevoked requests from stockholders holding in the aggregate less than the requisite number of shares entitling the stockholders to request the calling of a special meeting, the Board, in its discretion, may cancel the special meeting. Business transacted at a special meeting requested by stockholders shall be limited to the purpose(s) stated in the request for meeting, provided, however, that the Board shall have the authority in its discretion to submit additional matters to the stockholders, and to cause other business to be transacted, at any special meeting requested by stockholders.

Section 4. Adjournments. Any meeting of the stockholders, annual or special, may be adjourned from time to time to reconvene at the same or some other place, if any, and notice need not be given of any such adjourned meeting if the time, place, if any, thereof and the means of remote communication, if any, are announced at the meeting at which the adjournment is taken. At the adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting. If the adjournment is for more than 30 days, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date is fixed for stockholders entitled to vote at the adjourned meeting, the Board shall fix a new record date for notice of the adjourned meeting and shall give notice of the adjourned meeting to each stockholder of record entitled to vote at the adjourned meeting as of the record date fixed for notice of the adjourned meeting.

Section 5. Quorum. At each meeting of the stockholders, a majority of the shares of the Corporation entitled to vote at the meeting, present in person or represented by proxy, shall constitute a quorum. If, however, such quorum shall not be present or represented at any meeting of the stockholders, the stockholders entitled to vote thereat, present in person or represented by proxy, shall have power, by the affirmative vote of a majority in voting power thereof, to adjourn the meeting from time to time, in the manner provided in Section 4, until a quorum shall be present or represented. A quorum, once established, shall not be broken by the subsequent withdrawal of enough votes to leave less than a quorum. At any such adjourned meeting at which there is a quorum, any business may be transacted that might have been transacted at the meeting originally called.

Section 6. Voting; Proxies. The election of directors shall be by written ballot and shall be decided by a plurality of the votes cast at a meeting of the stockholders by the holders of stock entitled to vote in the election. Except as otherwise provided by law, the Certificate of Incorporation or these Bylaws, all action taken by the holders of a majority of the voting power represented at any meeting at which a quorum is present shall be valid and binding upon the Corporation. Each stockholder entitled to vote at a meeting of stockholders or to express consent to corporate action in writing without a meeting may authorize another person(s) to act for such stockholder by proxy, but no such proxy shall be voted or acted upon after three years from its date. A proxy shall be irrevocable if it states that it is irrevocable and if, and only as long as, it is coupled with an interest sufficient in law to support an irrevocable power. A stockholder may revoke any proxy which is not irrevocable by attending the meeting and voting in person or by delivering to the Secretary of the Corporation a revocation of the proxy or a new proxy bearing a later date. Voting at meetings of stockholders need not be by written hallot

Section 7. Notice. Written notice of any meeting of stockholders shall be given to each stockholder entitled to vote thereat, not less than ten nor more than 60 days prior to the meeting. If mailed, such notice shall be deemed to be given when deposited in the mail, postage prepaid, directed to the stockholder at his or her address as it appears on the records of the Corporation.

Section 8. Inspectors of Election. The Corporation may, in advance of, or at, any meeting of stockholders, appoint one or more inspectors of election to act at the meeting and make a written report thereof. The Corporation may designate one or more persons as alternate inspectors to replace any inspector who fails to act. In the event that no inspector so appointed or designated is able to act at a meeting of stockholders, the person presiding at the meeting shall appoint one or more inspectors to act at the meeting. The inspector(s) so appointed or designated shall (i) ascertain the number of shares of capital stock of the Corporation outstanding and the voting power of each such share, (ii) determine the shares of capital stock of the Corporation present or represented at the meeting and the validity of proxies and ballots, (iii) count all votes and ballots, (iv) determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspectors, and (v) certify their determination of the number of shares of capital stock of the Corporation present or represented at the meeting and such inspectors' count of all votes and ballots. Such certification shall specify such other information as may be required by law. In determining the validity and counting of proxies and ballots cast at any meeting of stockholders of the Corporation, the inspectors may consider such information as is permitted by applicable law. No person who is a candidate for an office at an election may serve as an inspector at such election.

Section 9. List of Stockholders Entitled to Vote. At least ten days before every meeting of the stockholders a complete list of the stockholders entitled to vote at said meeting, arranged in alphabetical order, with the address of each, and the number of shares held by each, shall be prepared by the Secretary. Such list shall be open to the examination of any stockholder for any purpose germane to the meeting, during ordinary business hours at the Corporation's headquarters or on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, and shall be produced and kept at the time and place of meeting during the whole time thereof and subject to the inspection of any stockholder who may be present. The original or duplicate stock ledger shall be provided at the time and place of each meeting and shall be the only evidence as to who are the stockholders entitled to examine the list of stockholders or to vote in person or by proxy at such meeting.

Section 10. Organization. The Chairman, or in his absence or if there is no Chairman, the Chief Executive Officer, or in his absence or if there is no Chief Executive Officer, the President, or in his absence or if there is no President, any Vice President, shall preside at meetings of the stockholders. The Secretary shall act as secretary of the meeting, or in his absence, an Assistant Secretary shall act, or if neither is present, then the presiding officer shall appoint a person to act as secretary of the meeting.

Section 11. <u>Conduct of Meetings</u>. The Board may adopt by resolution such rules and regulations for the conduct of the meeting of the stockholders as it shall deem appropriate. Except to the extent inconsistent with such rules and regulations as adopted by the Board, the Chairman of any meeting of the stockholders shall have the right and authority to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such Chairman, are appropriate for the proper conduct of the meeting.

Section 12. Fixing Date for Determination of Stockholders of Record In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of the stockholders or any adjournment thereof, or entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board and which record date: (1) in the case of determination of stockholders entitled to vote at any meeting of stockholders or adjournment thereof, shall, unless otherwise required by law, not be more than 60 nor less than ten days before the date of such meeting; and (2) in the case of any other action, shall not be more than 60 days prior to such other action. If no record date is fixed: (a) the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the date next preceding the day on which the meeting is held; and (b) the record date for determining stockholders for any other purpose shall be at the close of business on the day on which the of Board adopts the resolution relating thereto. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board may fix a new record date for the adjourned meeting.

Section 13. Nominations and Stockholder Business.

(a) Nominations of persons for election to the Board of the Corporation and the proposal of business other than nominations to be considered by the stockholders may be made at an annual meeting of stockholders (i) by or at the direction of the Board, or (ii) by any stockholder of the Corporation who is a stockholder of record at the time of giving the notice provided for in paragraphs (b) and (c) of this Section 13, who is entitled to vote at the meeting and who complied with the notice procedures set forth in this Section 13.

(b) In order to assure that stockholders and the Corporation have a reasonable opportunity to consider nominations and other business proposed to be brought before a meeting of stockholders and to allow for full information to be distributed to stockholders, a stockholder properly may bring nominations or other business before an annual meeting of stockholders only if the stockholder shall have given timely notice thereof in writing to the Secretary of the Corporation setting forth the information required by paragraph (c) of this Section 13, and such other business must be a proper subject for stockholder action under the Delaware General Corporation Law. To be timely, a stockholder's notice shall be delivered to the Secretary at the principal executive offices of the Corporation (if delivered by electronic mail or facsimile, the stockholder's notice shall be directed to the Secretary at the electronic mail address or facsimile number, as the case may be, specified in the Corporation's most recent proxy statement) not later than the close of business on the 45th day nor earlier than 120 days prior to the anniversary of the date on which the Corporation filed its definitive proxy materials (regardless of whether or not thereafter revised or supplemented) with the Securities and Exchange Commission for the prior year's annual meeting of stockholders; provided, however, that in the event that the date of the annual meeting is advanced by more than 30 days or delayed (other than as a result of adjournment) by more than 30 days from the anniversary of the previous year's annual meeting, notice by the stockholder to be timely must be delivered not later than the close of business on the later of the 60th day prior to such annual meeting or the tenth day following the day on which public announcement of the date of such meeting is first made. In no event shall the public announcement of an adjournment or postponement of an annual meeting commence a new time period (or extend any time period) for the giving of a stock

(c) Such stockholder's notice shall set forth and include:

- (i) as to each person whom the stockholder proposes to nominate for election or reelection as a director (A) all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors in an election contest, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), (B) such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected and (C) the information necessary for the Board to determine whether such proposed nominee qualifies as an independent director under any applicable stock exchange listing standards;
- (ii) as to any other business that the stockholder proposes to bring before the meeting, (A) a brief description of the business desired to be brought before the meeting, (B) the reasons for conducting such business at the meeting, and (C) a description of any substantial interest in such business of such stockholder and the beneficial owner (within the meaning of Item 5 of Schedule 14A under the Exchange Act), if any, on whose behalf the proposal is made;
- (iii) as to the stockholder giving the notice and the beneficial owner, if any, on whose behalf the nomination or proposal is made (A) the name and address of such stockholder, as they appear on the Corporation's books, and the name and address of such beneficial owner, (B) the class and number of shares of the Corporation which are owned of record by such stockholder and such beneficial owner as of the date of the notice, and a representation that the stockholder will notify the Corporation in writing within five business days after the record date for such meeting of the class and number of shares of the Corporation owned of record by the stockholder and such beneficial owner as of the record date for the meeting, and (C) a representation that the stockholder intends to be present in person or by proxy at the meeting to propose such nomination or other business;

(iv) as to the stockholder giving the notice or, if the notice is given on behalf of a beneficial owner on whose behalf the nomination or proposal is made, as to such beneficial owner (A) the class and number of shares of the Corporation which are beneficially owned by such stockholder or beneficial owner as of the date of the notice and by each associate of the stockholder or beneficial owner as of the date of the notice, and a representation that the stockholder will notify the Corporation in writing within five business days after the record date for such meeting of the class and number of shares of the Corporation beneficially owned by such stockholder or beneficial owner and by each such associate as of the record date for the meeting, (B) a description of any agreement, arrangement or understanding (whether or not in writing) with respect to the nomination or other business between or among such stockholder or beneficial owner and any other person, including without limitation any agreements that would be required to be described or reported pursuant to Item 5 or Item 6 of Schedule 13D (regardless of whether the requirement to file a Schedule 13D is applicable to the stockholder or beneficial owner) and a representation that the stockholder will notify the Corporation in writing within five business days after the record date for such meeting of any such agreement, arrangement or understanding in effect as of the record date for the meeting, (C) a description of any agreement, arrangement or understanding (including any derivative or short positions, profit interests, options, hedging transactions, and borrowed or loaned shares, regardless of whether settled in shares or cash) that has been entered into as of the date of the stockholder's notice by, or on behalf of, such stockholder or beneficial owner, the effect or intent of which is to mitigate loss, manage risk or benefit from changes in the share price of any class of the Corporation's capital stock, or increase or decrease the voting power of the stockholder or beneficial owner with respect to shares of stock of the Corporation, including the notional number of shares that are the subject of such agreement, arrangement or understanding, and a representation that the stockholder will notify the Corporation in writing within five business days after the record date for such meeting of any such agreement, arrangement or understanding in effect as of the record date for the meeting, (D) a description of any agreement, arrangement or understanding (whether or not in writing) between or among such stockholder or beneficial owner and any other person relating to acquiring, holding, voting or disposing of any shares of stock of the Corporation, including the number of shares that are the subject of such agreement, arrangement or understanding, and a representation that the stockholder will notify the Corporation in writing within five (5) business days after the record date for such meeting of any such agreement, arrangement or understanding in effect as of the record date for the meeting, and (E) a representation as to whether the stockholder or beneficial owner will engage in a solicitation with respect to such nomination or proposal and, if so, the name of each participant (as defined in Item 4 of Schedule 14A under the Exchange Act) in such solicitation and whether such person or group intends to deliver a proxy statement and/or form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve or adopt the business to be proposed (in person or by proxy) by the stockholder; and

(v) as to the stockholder giving the notice and the beneficial owners if any on whose behalf the nomination or proposal is made, such stockholder's and beneficial owner's written consent to the public disclosure of information provided pursuant to paragraph (c) of this Section 13.

(d) The requirements of this Section 13 shall not apply to a stockholder if the stockholder has notified the Corporation of his or her intention to present a non-binding (precatory) stockholder proposal at an annual meeting pursuant to and in compliance with Rule 14a-8 under the Exchange Act. With respect to any other matter proposed to be presented pursuant to and in compliance with Rule 14a-8, (i) the notice required by this Section 13 shall be considered timely if delivered within the time period specified in Rule 14a-8(e), and (ii) the person proposing to have such matter presented at the meeting shall provide the information required by paragraph (c) of this Section 13, provided that the information required by paragraph (c)(iii) of this Section 13 may be satisfied by providing the information required pursuant to Rule 14a-8(b).

- (e) Nominations of persons for election to the Board may be made at a special meeting of stockholders at which directors are to be elected pursuant to the Corporation's notice of meeting (a) by or at the direction of the Board or (b) by any stockholder of the Corporation who is a stockholder of record at the time of giving of notice provided for in this Section 13, who is entitled to vote at the meeting and who complies with the notice procedures set forth in this Section 13. Nominations by stockholders of persons for election to the Board may be made at such a special meeting of stockholders if the nominees are named in the special meeting request delivered or, at any special meeting called, if the stockholder's notice required by this Section 13 is delivered to the Secretary at the principal executive offices of the Corporation (if delivered by electronic mail or facsimile, the stockholder's notice shall be directed to the Secretary at the electronic mail address or facsimile number, as the case may be, specified in the Corporation's most recent proxy statement) not earlier than the close of business on the 120th day prior to such special meeting and not later than the close of business on the later of the 60th day prior to such special meeting or the tenth day following the day on which public announcement is first made of the date of the special meeting and such notice sets forth the information required by paragraph (c) of this Section 13. In no event shall the public announcement of an adjournment or postponement of a special meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above.
- (f) Only those persons who are nominated in accordance with the procedures set forth in this Section 13 shall be eligible for election as directors at any meeting of stockholders. Only such other business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with the procedures set forth in this Section 13. The Chairman of the Board or Secretary may, if the facts warrant, determine that a notice received by the Corporation relating to a nomination proposed to be made or an item of business proposed to be introduced at a meeting of stockholders does not satisfy the requirements of this Section 13 (including if the stockholder does not provide the information required under Section 13(c)(iii)(B) and 13(c)(iv)(A)-(C) to the Corporation within five business days following the record date for the meeting), and if it be so determined, shall so declare and any such nomination or other business shall not be introduced at such meeting of stockholders, notwithstanding that proxies in respect of such matters may have been received. The chairman of the meeting shall have the power and duty to determine whether a nomination or any other business brought before the meeting was made in accordance with the procedures set forth in this Section 13, and, if any nomination or other business is not in compliance with this Section 13 (including if the stockholder does not provide the information required under Section 13(c)(iii)(B) and 13(c)(iv)(A)-(C) to the Corporation within five business days following the record date for the meeting), to declare that such defective nomination or proposal shall be disregarded, notwithstanding that proxies in respect of such matters may have been received. Notwithstanding the foregoing provisions of this Section 13, if the stockholder (or a qualified representative of the stockholder) is not present at the meeting of stockholders to make a nomination or propose such business, such nomination shall be disregarded and such proposed business shall not be transacted, notwithstanding that proxies in respect of such vote have been received by the Corporation. For purposes of this Section 13, to be considered a qualified representative of the stockholder, a person must be a duly authorized officer, manager or partner of such stockholder or authorized by a writing executed by such stockholder (or a reliable reproduction or electronic transmission of the writing) delivered to the Corporation prior to the making of such nomination or proposal at such meeting by such stockholder stating that such person is authorized to act for such stockholder as proxy at the meeting of stockholders.
- (g) For purposes of this Section 13, "public announcement" shall mean disclosure in a press release or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Section 13, 14 or 15(d) of the Exchange Act. For purposes of paragraph (c)(iv)(A) of this Section 13, the term "associate" shall have the meaning set forth in Rule 14a-1(a) under the Exchange Act and shares shall be treated as "beneficially owned" by a person if the person (i) beneficially owns such shares, directly or indirectly, for purposes of Section 13(d) of the Exchange Act and Regulations 13D and 13G thereunder, or (ii) has or shares pursuant to any agreement, arrangement or understanding (whether or not in writing) (A) the right to acquire such shares (whether such right is exercisable immediately or only after the passage of time or the fulfillment of a condition or both), (B) the right to vote such shares, alone or in concert with others, and/or (C) investment power with respect to such shares, including the power to dispose of, or to direct the disposition of, such shares.
- (h) Notwithstanding the foregoing provisions of this Section 13, a stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Section 13.

ARTICLE II BOARD OF DIRECTORS

- Section 1. Number; Election; Term. The number of directors which shall constitute the whole Board shall not be less than three (3), nor more than twelve (12), the exact number within said limits to be fixed from time to time solely by resolution of the Board, acting by the vote of not less than a majority of the directors then in office. Should the death, resignation or other removal of any non-employee director result in the failure of the requirement set forth in the preceding sentence to be met, such requirement shall not apply during the time of the vacancy caused by the death, resignation or removal of any such non-employee director. The remaining directors of the Corporation shall cause any such vacancy to be filled in accordance with these bylaws within a reasonable period of time. At the annual meeting or a special meeting at which directors are to be elected in accordance with the Corporation's notice of meeting, directors shall be elected in accordance with the requirements of these bylaws.
- Section 2. <u>Place of Meetings</u>; <u>Records</u>. The directors may hold their meetings and keep the books of the Corporation outside of the State of Delaware at such places as they may from time to time determine.
- Section 3. <u>Powers</u>. The powers of the Corporation shall be exercised, its business conducted and its property controlled by or under the direction of the Board.
- Section 4. <u>Vacancies</u>. If the office of any director becomes vacant for any reason or any new directorship is created by any increase in the authorized number of directors, a majority of the directors then in office, although less than a quorum, may choose a successor or successors or fill the newly created directorship. Any director so chosen shall hold office until the next annual election and until his successor shall be elected and qualified.

Section 5. Resignation and Removals.

- (a) Any director may resign at any time by delivering his written resignation to the Secretary, such resignation to specify whether it will be effective at a particular time, upon receipt by the Secretary or at the pleasure of the Board. If no such specification is made, it shall be deemed effective at the pleasure of the Board. When one or more directors shall resign from the Board, effective at a future date, a majority of the directors then in office, including those who have so resigned, shall have power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each director so chosen shall hold office for the unexpired portion of the term of the director whose place shall be vacated and until his successor shall have been duly elected and qualified.
- (b) Except as provided in Section 141 of the Delaware General Corporation Law, at a special meeting of stockholders called for the purpose in the manner hereinabove provided, the Board, or any individual director, may be removed from office, with or without cause, and a new director or directors elected by a vote of stockholders holding a majority of the outstanding shares entitled to vote at an election of directors.
- Section 6. Regular Meetings. Regular meetings of the Board may be held without notice at such time and place either within or without the State of Delaware as shall from time to time be determined by the Board.
- Section 7. Special Meetings. Special meetings of the Board may be called by the Chair of the Board, a Vice-Chair of the Board or the President and on the written request of any two directors, by the Secretary. Written notice of the time and place of all special meetings of the Board shall be delivered to each director at least 24 hours before the start of the meeting, or if sent by first class mail, at least 72 hours before the start of the meeting. Notice of any meeting may be waived in writing at any time before or after the meeting and will be waived by any director by attendance thereat.

Section 8. <u>Telephonic Meetings</u>. To the extent permitted by law, members of the Board or any committee thereof may participate in a meeting of such body through the use of conference telephone or similar communications equipment by means of which all persons participating in the meeting can hear each other, and participants in a meeting pursuant to this Section 8 shall constitute presence in person at such meeting.

Section 9. Quorum. The presence of a majority of the Board shall be necessary and sufficient to constitute a quorum for the transaction of business at any meeting of the Board.

Section 10. Committees of the Board. The Board may designate one or more committees, each committee to consist of one or more of the directors of the Corporation. The Board may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. Unless the Board provides otherwise, at all meetings of such committee, a majority of the then authorized members of the committee shall constitute a quorum for the transaction of business, and the vote of a majority of the members of the committee present at any meeting at which there is a quorum shall be the act of the committee. Each committee shall keep regular minutes of its meetings. Unless the Board provides otherwise, each committee designated by the Board may make, alter and repeal rules and procedures for the conduct of its business. In the absence of such rules and procedures each committee shall conduct its business in the same manner as the Board conducts its business pursuant to this Article II.

Section 11. <u>Action Without Meetings</u>. Any action required or permitted to be taken at any meeting of the Board or any committee designated by such Board may be taken without a meeting, if all members of the Board or committee consent thereto in writing or by electronic transmission and the writing or writings or electronic transmission or transmissions are filed with the minutes of the proceedings of the Board or committee.

Section 12. Chairman of the Board of Directors and Other Board Positions Designated

- (a) The Board shall elect one of its members to be Chairman of the Board and may elect one or more Vice Chairman and, a Chairman of each committee of the Board, and such other positions as the Board may deem expedient. The Board shall fill any vacancy in the position of Chairman of the Board at such time and in such manner as the Board shall determine.
- (b) The Chairman of the Board shall preside at all meetings of the stockholders and the Board, unless the Board determines otherwise. The Chairman of the Board shall perform such other duties and have such other powers as the Board shall designate from time to time.
- (c) Any Vice-Chairman shall preside at meetings of the Board in the absence of the Chairman, unless the Board determines otherwise. The Vice Chair shall have such authority as specified by the Chairman, the Board or these Bylaws.

ARTICLE III OFFICERS

- Section. 1 <u>Positions and Election</u>. The officers of the Corporation shall be elected annually by the Board and shall include a President, a Vice President, a Treasurer and a Secretary. The Board, in its discretion, may also elect a Chairman (who must be a director), one or more Vice Chairman (who must be directors) and one or more Vice Presidents, Assistant Treasurers, Assistant Secretaries and other officers. Any two or more offices may be held by the same person.
- Section 2. Term. Each officer of the Corporation shall hold office until such officer's successor is elected and qualified or until such officer's earlier death, resignation or removal. Any officer elected or appointed by the Board may be removed by the Board at any time with or without cause by the majority vote of the members of the Board then in office. The removal of an officer shall be without prejudice to his or her contract rights, if any. The election or appointment of an officer shall not of itself create contract rights. Any officer of the Corporation may resign at any time by giving written notice of his or her resignation to the President or the Secretary. Any such resignation shall take effect at the time specified therein or, if the time when it shall become effective shall not be specified therein, immediately upon its receipt. Unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective. Should any vacancy occur among the officers, the position shall be filled for the unexpired portion of the term by appointment made by the Board.
- Section 3. <u>The President</u>. The President shall have general supervision over the business of the Corporation and other duties incident to the office of President, and any other duties as may be from time to time assigned to the president by the Board and subject to the control of the Board in each case.
- Section 4. Vice Presidents. Each Vice President shall have such powers and perform such duties as may be assigned to him or her from time to time by the Chairman of the Board or the President.
- Section 5. Secretary. The Secretary shall attend all sessions of the Board and all meetings of the stockholders and record all votes and the minutes of all proceedings in a book to be kept for that purpose, and shall perform like duties for committees when required. He or she shall give, or cause to be given, notice of all meetings of the stockholders and meetings of the Board, and shall perform such other duties as may be prescribed by the Board or the President. The Secretary shall keep in safe custody the seal of the Corporation and have authority to affix the seal to all documents requiring it and attest to the same.
- Section 6. Treasurer. The Treasurer shall have the custody of the corporate funds and securities, except as otherwise provided by the Board, and shall keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation and shall deposit all moneys and other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the Board. The Treasurer shall disburse the funds of the Corporation as may be ordered by the Board, taking proper vouchers for such disbursements, and shall render to the President and the directors, at the regular meetings of the Board, or whenever they may require it, an account of all his or her transactions as Treasurer and of the financial condition of the Corporation.
- Section 7. <u>Duties of Officers May be Delegated.</u> In case any officer is absent, or for any other reason that the Board may deem sufficient, the President or the Board may delegate for the time being the powers or duties of such officer to any other officer or to any director.

<u>ARTICLE IV</u> STOCK CERTIFICATES AND THEIR TRANSFER

Section 1. Certificates Representing Shares. The shares of stock of the Corporation shall be represented by certificates; provided that the Board may provide by resolution or resolutions that some or all of any class or series shall be uncertificated shares that may be evidenced by a book-entry system maintained by the registrar of such stock. If shares are represented by certificates, such certificates shall be in the form, other than bearer form, approved by the Board. The certificates representing shares of stock of each class shall be signed by, or in the name of, the Corporation by the Chairman, any Vice Chairman, the President or any Vice President, and by the Secretary, any Assistant Secretary, the Treasurer or any Assistant Treasurer. Any or all such signatures may be facsimiles. Although any officer, transfer agent or registrar whose manual or facsimile signature is affixed to such a certificate ceases to be such officer, transfer agent or registrar before such certificate has been issued, it may nevertheless be issued by the Corporation with the same effect as if such officer, transfer agent or registrar were still such at the date of its issue.

Section 2. <u>Transfers of Stock</u>. Transfers of stock shall be made upon the books of the Corporation: (1) upon presentation of the certificates by the registered holder in person or by duly authorized attorney, or upon presentation of proper evidence of succession, assignment or authority to transfer the stock, and upon surrender of the appropriate certificate(s), or (2) in the case of uncertificated shares, upon receipt of proper transfer instructions from the registered owner of such uncertificated shares, or from a duly authorized attorney or from an individual presenting proper evidence of succession, assignment or authority to transfer the stock.

Section 3. Lost, Stolen or Destroyed Certificates. The Board may direct a new certificate or uncertificated shares to be issued in place of any certificate theretofore issued by the Corporation alleged to have been lost, stolen or destroyed upon the making of an affidavit of that fact by the owner of the allegedly lost, stolen or destroyed certificate. When authorizing such issue of a new certificate or uncertificated shares, the Board may, in its discretion and as a condition precedent to the issuance thereof, require the owner of the lost, stolen or destroyed certificate, or the owner's legal representative to give the Corporation a bond sufficient to indemnify it against any claim that may be made against the Corporation with respect to the certificate alleged to have been lost, stolen or destroyed or the issuance of such new certificate or uncertificated shares.

Section 4. Holder of Record. The Corporation shall be entitled to treat the holder of record of any share or shares of stock as the holder in fact thereof and accordingly shall not be bound to recognize any equitable or other claim to or interest in such share on the part of any other person whether or not it shall have express or other notice thereof, save as expressly provided by the laws of the State of Delaware.

ARTICLE V INDEMNIFICATION

Each person who was or is a party or is threatened to be made a party to or is involved (as a party, witness, or otherwise), in any threatened, pending, or completed action, suit, arbitration, alternative dispute mechanism, inquiry, administrative or legislative hearing, investigation or any other actual, threatened or completed proceeding, including any and all appeals, whether civil, criminal, administrative, or investigative (hereinafter a "Proceeding"), by reason of the fact that he or she, or a person of whom he or she is the legal representative, is or was a director, officer, employee, or agent of the Corporation (including service with respect to employee benefit plans) or is or was serving at the request of the Corporation as a director, officer, employee, or agent of another corporation or of a partnership, joint venture, trust, or other enterprise, whether the basis of the Proceeding is alleged action in an official capacity as a director, officer, employee, or agent or in any other capacity while serving as a director, officer, employee, or agent (hereafter an "Agent"), shall be indemnified and held harmless by the Corporation to the fullest extent authorized by the Delaware General Corporation Law, as the same exists or may hereafter be amended or interpreted (but, in the case of any such amendment or interpretation, only to the extent that such amendment or interpretation permits the Corporation to provide broader indemnification rights than were permitted prior thereto) against all expenses, liability, and loss (including attorneys' fees, judgments, fines, ERISA excise taxes or penalties, and amounts paid or to be paid in settlement, and any interest, assessments, or other charges imposed thereon, and any federal, state, local, or foreign taxes imposed on any Agent as a result of the actual or deemed receipt of any payments under this Article) reasonably incurred or suffered by such person in connection with investigating, defending, being a witness in, or participating in (including on appeal), or preparing for any of the foregoing in, any Proceeding (hereinafter "Expenses"); provided, however, the Corporation shall indemnify any Agent seeking indemnification in connection with a Proceeding (or part thereof) initiated by such person only if the Proceeding (or part thereof) was authorized by the Board of the Corporation. The right to indemnification conferred in this Article shall be a contract right.

ARTICLE VI GENERAL PROVISIONS

- Section 1. Offices. The address of the registered office of the Corporation in the State of Delaware shall be at c/o The Corporation Trust Company, Corporation Trust Center, 1209 Orange Street, Wilmington, Delaware 19801. The Corporation may have other offices, both within and without the State of Delaware, as the Board of Directors of from time to time shall determine or the business of the Corporation may require.
- Section 2. <u>Seal</u>. The seal of the Corporation shall be in such form as shall be approved by the Board. The seal may be used by causing it or a facsimile thereof to be impressed or affixed or reproduced or otherwise, as may be prescribed by law or custom or by the Board.
 - Section 3. Fiscal Year. The fiscal year of the Corporation shall start on January 1 and end on December 31 of each year.
- Section 4. Checks, Notes, Drafts, Etc. All checks, notes, drafts or other orders for the payment of money of the Corporation shall be signed, endorsed or accepted in the name of the Corporation by such officers, officers, person or persons as from time to time may be designated by the Board or by an officer or officers authorized by the Board to make such designation.
- Section 5. Notice. Whenever notice is required to be given by law, the Certificate of Incorporation or these Bylaws, a written waiver signed by the person entitled to notice, whether before or after the time stated therein, shall be deemed equivalent to notice. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting except when the person attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting, is not lawfully called or convened.
- Section 6. Amendments. Unless otherwise provided in the Certificate of Incorporation, these Bylaws may be repealed, altered or amended or new Bylaws adopted at any meeting of the stockholders, either annual or special, by the affirmative vote of a majority of the stock entitled to vote at such meeting. The Board shall also have the authority to repeal, alter or amend these Bylaws or adopt new Bylaws (including, without limitation, the amendment of any Bylaws setting forth the number of directors who shall constitute the whole Board) by unanimous written consent or at any annual, regular, or special meeting by the affirmative vote of a majority of the whole number of directors, subject to the power of the stockholders to change or repeal such Bylaws and provided that the Board shall not make or alter any Bylaws fixing the qualifications, classifications, term of office or compensation of directors.
- Section 7. Form of Records. Any records maintained by the Corporation in the regular course of its business, including its stock ledger, books of account, and minutes books, may be kept on, or by means of, or be in the form of, any information storage device, or method provided that the records so kept can be converted into clearly legible paper form within a reasonable time. The Corporation shall so convert any records so kept upon the request of any person entitled to inspect the same.

BIRCHTECH CORP.

AMENDED AND RESTATED 2014 EQUITY INCENTIVE PLAN

BIRCHTECH CORP.

(formerly Midwest Energy Emissions Corp.) AMENDED AND RESTATED 2014 EQUITY INCENTIVE PLAN Adopted July 3, 2023; Amended October 29, 2024

Adopted July 5, 2025, Amended October 25, 2024

1. **Purpose.** The purpose of the Birchtech Corp. Amended and Restated 2014 Equity Incentive Plan is to provide officers, other employees and directors of, and consultants to, Birchtech Corp. or any of its Subsidiaries an incentive (a) to enter into and remain in the service of the Company or its Subsidiaries, (b) to enhance the long-term performance of the Company and its Subsidiaries, and (c) to acquire a proprietary interest in the success of the Company and its Subsidiaries.

2. Definitions.

2.1 **Definitions in the Plan.** Wherever the following capitalized terms are used in the Plan, they shall have the meanings specified below:

"Award" means an award of a Stock Option, Stock Appreciation Right, Restricted Stock Unit Award, Performance Share Award, Performance Unit Award or Stock Award granted under the Plan.

"Award Agreement" means a written or electronic agreement entered into between the Company and a Participant setting forth the terms and conditions of an Award granted to a Participant.

"Board" means the Board of Directors of the Company.

"Cause" means termination of Participant's employment for "cause" as defined in any employment or severance agreement the Participant may have with the Company or a Subsidiary or, if no such agreement exists, unless otherwise provided in a particular Award Agreement, "cause" means (a) conviction or pleading guilty or no contest to any crime (whether or not involving the Company or any of its Subsidiaries) constituting a felony or involving fraud, dishonesty or moral turpitude in the jurisdiction involved; (b) engaging in any act which, in each case, subjects, or if generally known would subject, the Company or any of its Subsidiaries to public ridicule or embarrassment; (c) material violation of the Company's or any of its Subsidiaries' policies, including, without limitation, those relating to sexual harassment or the disclosure or misuse of confidential information; (d) serious neglect or misconduct in the performance of the Participant's duties for the Company or any of its Subsidiaries or willful or repeated failure or refusal to perform such duties; in each case as determined by the Committee, which determination will be final, binding and conclusive.

"Change in Control" means: (A) the acquisition by any person or group (as that term is defined in Section 13 of the Securities Exchange Act of 1934, as amended) of more than 50% of the outstanding Common Stock, (B) a consolidation or merger of the Company with another entity, unless immediately after the transaction, at least 50% in voting power of the outstanding shares or other equity interests in the surviving entity or its ultimate parent entity are owned by persons who, immediately before the transaction were shareholders of the Company, or (C) the consummation of the sale or disposition by the Company or all or substantially all of the Company's assets. Notwithstanding the foregoing, no event or condition will constitute a Change in Control to the extent (but only to the extent) that, if it were a Change of Control, a 20% tax would be imposed under Section 409A of the Code.

- "Code" means the Internal Revenue Code of 1986, as amended.
- "Common Stock" means the Company's common stock, par value \$.001 per share.
- "Committee" means the Compensation Committee of the Board, or another committee of the Board appointed by the Board to administer the Plan.
 - "Company" means Birchtech Corp., a Delaware corporation.
- "Date of Grant" means the date on which an Award under the Plan is made by the Committee, or such later date as the Committee may specify to be the effective date of an Award.
- "Disability" means termination of Participant's employment for "disability" as defined in any employment or severance agreement the Participant may have with the Company or a Subsidiary or, if no such agreement exists, unless otherwise provided in a particular Award Agreement, a Participant being considered "disabled" within the meaning of Section 409A(a)(2)(C) of the Code, except that no circumstance or condition will constitute a Disability to the extent (but only to the extent) that, if it were, a 20% tax would be imposed under Section 409A of the Code.
- "Eligible Person" means any person who is an officer, employee of or consultant to the Company or any Subsidiary (including any key employee of an entity that develops products that are intended to be published or distributed by the Company or a Subsidiary) or any person to whom an offer of employment with the Company or any Subsidiary is extended, as determined by the Committee, or any person who is a Non-Employee Director.
 - "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- "Fair Market Value" of a share of Common Stock as of a given date shall be, if the Common Stock is listed on any established stock exchange or traded on any established market, the closing price of the Common Stock on the trading day immediately prior to the applicable date or, if there is no closing price on the trading day immediately prior to the applicable date, the closing price on the last preceding date for which such quotation exists. In any situation not covered by the above, the Fair Market Value of a share of Common Stock on any day will be determined in good faith by the Committee and in a manner that complies with Sections 409A and 422 of the Code.
- "Good Reason" means termination of Participant's employment for "good reason" as defined in any employment or severance agreement the Participant may have with the Company or a Subsidiary or, if no such agreement exists, unless otherwise provided in a particular Award Agreement, "good reason" means (A) a material reduction of the Participant's authority or responsibilities, (B) the assignment to the Participant of duties materially inconsistent with the Participant's position with the Company or a Subsidiary; (C) a reduction in Participant's annual salary or an alteration of the formula by which the Participant's annual bonus is calculated that is likely to lead to a reduction in the Participant's annual compensation; or (D) the relocation of the Participant's office by more than 100 miles, except, in each case, to the extent consented to by the Participant in writing.

- "Incentive Stock Option" means a Stock Option granted under Section 6 hereof that is intended to meet the requirements of Section 422 of the Code and the regulations thereunder.
 - "Non-Employee Director" means any member of the Board who is not an employee of the Company.
 - "Nonqualified Stock Option" means a Stock Option granted under Section 6 hereof that is not an Incentive Stock Option.
 - "Participant" means any Eligible Person who holds an outstanding Award under the Plan.
- "Performance Share Award" means a contractual right granted to an Eligible Person under Section 10 hereof representing unit interests equal in value to a share of Common Stock that is forfeitable until the achievement of pre-established performance objectives over a performance period.
- "Performance Unit Award" means a contractual right granted to an Eligible Person under Section 10 hereof representing unit interests equal to a pre-determined dollar amount that is forfeitable until the achievement of pre-established performance objectives over a performance period.
 - "Plan" means the Birchtech Corp. Amended and Restated 2014 Equity Incentive Plan as set forth herein, as amended from time to time.
- "Restricted Stock Unit Award" means a contractual right granted to an Eligible Person under Section 8 hereof representing unit interests equal in value to a share of Common Stock to be paid or distributed at such times, and subject to such conditions, as set forth in the Plan and the applicable Award Agreement.
- "Retirement" means termination of Participant's employment by reason of "retirement" as defined in any employment or severance agreement the Participant may have with the Company or a Subsidiary or, if no such agreement exists, unless otherwise provided in a particular Award Agreement, "retirement" means a termination of the Participant's Service after the Participant reaches the age of 65.
- "Service" means a Participant's employment or service with the Company or any Subsidiary or a Participant's service as a Non-Employee Director with the Company, as applicable.
- "Stock Award" means a grant of shares of Common Stock, or securities that are convertible into Common Stock, or other equity-based Award, to an Eligible Person under Section 11 hereof.
- "Stock Appreciation Right" means a contractual right granted to an Eligible Person under Section 7 hereof entitling such Eligible Person to receive a payment, representing the difference between the base price per share of the right and the Fair Market Value of a share of Common Stock, at such time, and subject to such conditions, as are set forth in the Plan and the applicable Award Agreement.

"Stock Option" means a contractual right granted to an Eligible Person under Section 6 hereof to purchase shares of Common Stock at such time and price, and subject to such conditions, as are set forth in the Plan and the applicable Award Agreement.

"Subsidiary" means any corporation (other than the Company) in an unbroken chain of corporations beginning with the Company if, at the time as of which a determination is being made, each corporation other than the last corporation in the unbroken chain owns stock possessing 50% or more of the total combined voting power of all classes of stock in another corporation in the chain.

"TSX" means the Toronto Stock Exchange.

2.2 Definitions Incorporated by Reference. For the purposes of the Plan, the terms "employee", "consultant", "consultant company", "management company employee", "investor relations activities", "insiders" and "investor relations service providers" shall have the respective meanings ascribed thereto in the policies of the TSX.

3. Administration.

- 3.1 Committee Members. The Plan shall be administered by a Committee comprised of no fewer than two members of the Board. It is intended that each Committee member shall satisfy the requirements for (i) a "non-employee director" for purposes of Rule 16b-3 under the Exchange Act, to the extent desirable to qualify transactions hereunder as exempt under Rule 16b-3, and (ii) an "outside director" under Section 162(m) of the Code, unless the action taken pursuant to the Plan is not required to be taken by "outside directors" to qualify for tax deductibility under Section 162(m) of the Code, and (iii) if required by the rules of any securities exchange or market on which the Common Stock is listed, an "independent director" under those rules. No member of the Committee shall be liable for any action or determination made in good faith by the Committee with respect to the Plan or any Award. In the absence of a Committee, the Board will administer the Plan and all references to the "Committee" will be deemed to refer to the "Board".
- 3.2 Committee Authority. The Committee shall have such powers and authority as may be necessary or appropriate for the Committee to carry out its functions as described in the Plan. Subject to the express limitations of the Plan, the Committee shall have authority in its discretion to determine the Eligible Persons to whom, and the time or times at which, Awards may be granted, the number of shares, units or other rights subject to each Award, the exercise, base or purchase price of an Award (if any), the time or times at which an Award will become vested, exercisable or payable, the performance goals and other conditions affecting an Award, the duration of the Award, and all other terms of the Award. The Committee shall also have discretionary authority to interpret the Plan, to make factual determinations under the Plan, and to make all other determinations necessary or advisable for Plan administration, including, without limitation, to correct any defect, to supply any omission or to reconcile any inconsistency in the Plan or any Award Agreement. The Committee may prescribe, amend, and rescind rules and regulations relating to the Plan. The Committee's determinations under the Plan need not be uniform and may be made by the Committee selectively among Participants and Eligible Persons, whether or not such persons are similarly situated. The Committee shall, in its discretion, consider such factors as it deems relevant in making its interpretations, determinations and actions under the Plan including, without limitation, the recommendations or advice of any officer or employee of the Company or such attorneys, consultants, accountants or other advisors as it may select. All interpretations, determinations and actions by the Committee shall be final, conclusive, and binding upon all parties.

- 3.3 Delegation of Authority. The Committee shall have the right, from time to time, to delegate to one or more officers of the Company the authority of the Committee to grant and determine the terms and conditions of Awards granted under the Plan, subject to the requirements of Section 157(c) of the Delaware General Corporation Law (or any successor provision) and such other limitations as the Committee shall determine. In no event shall any such delegation of authority be permitted with respect to Awards to any members of the Board or to any Eligible Person who is subject to Rule 16b-3 under the Exchange Act or Section 162(m) of the Code. The Committee shall also be permitted to delegate, to any appropriate officer or employee of the Company, responsibility for performing certain ministerial functions under the Plan. In the event that the Committee's authority is delegated to officers or employees in accordance with the foregoing, all provisions of the Plan relating to the Committee shall be interpreted in a manner consistent with the foregoing by treating any such reference as a reference to such officer or employee for such purpose. Any action undertaken in accordance with the Committee's delegation of authority hereunder shall have the same force and effect as if such action was undertaken directly by the Committee and shall be deemed for all purposes of the Plan to have been taken by the Committee.
- 3.4 Grants to Non-Employee Directors. Any Awards or formula for granting Awards under the Plan made to Non-Employee Directors shall be approved by the Board. With respect to awards to such directors, all rights, powers and authorities vested in the Committee under the Plan shall instead be exercised by the Board, and all provisions of the Plan relating to the Committee shall be interpreted in a manner consistent with the foregoing by treating any such reference as a reference to the Board for such purpose.

4. Shares Subject to the Plan.

4.1 Subject Shares. Prior to July 3, 2023, an aggregate of 218,638 shares of Common Stock have been issued under the Plan. Subject to adjustment pursuant to Section 4.4 hereof, the maximum aggregate number of shares of Common Stock that may be issued under the Plan after July 3, 2023 shall be four million seven hundred seventy-five thousand (4,775,000) shares. Shares of Common Stock issued under the Plan may be either authorized but unissued shares or shares held in the Company's treasury.

To the extent that any Award involving the issuance of shares of Common Stock is forfeited, cancelled, returned to the Company for failure to satisfy vesting requirements or other conditions of the Award, or otherwise terminates without an issuance of shares of Common Stock being made thereunder, the shares of Common Stock covered thereby will no longer be counted against the maximum share limitations and may again be made subject to Awards under the Plan pursuant to such limitations. In addition, awards that are settled in cash and not in shares of Common Stock shall not be counted against the maximum share limitations.

4.2 [intentionally deleted]

- 4.3 Limitations on Awards. Except where permitted by the policies of the TSX (including, without limitation, upon obtaining any disinterested stockholder approval required therefor) and so long as the shares of Common Stock are listed on the TSX:
- (a) the aggregate number of shares of Common Stock issuable to any one Participant (and any companies wholly owned by such Participant), pursuant to awards granted or issued in any twelve (12) month period under this Plan and all of the Company's other security-based compensation arrangements, shall not exceed five percent (5%) of the number of issued and outstanding shares of Common Stock at the date the award is granted;
- (b) the aggregate number of shares of Common Stock issuable to any one consultant, pursuant to awards granted or issued in any twelve (12) month period under this Plan and all of the Company's other security-based compensation arrangements, shall not exceed two percent (2%) of the number of issued and outstanding shares of Common Stock at the time the award is granted to the said consultant under this Plan;
- (c) the aggregate number of shares of Common Stock issuable to all investor relations service providers, pursuant to awards granted or issued in any twelve (12) month period under this Plan and all of the Company's other security-based compensation arrangements, shall not exceed two percent (2%) of the number of issued and outstanding shares of Common Stock at the time of any grant of an award under this Plan to an investor relations service provider;
- (d) the aggregate number of shares of Common Stock issuable to insiders of the Company (as a group), pursuant to awards granted or issued under this Plan and all of the Company's other security-based compensation arrangements, shall not exceed ten percent (10%) of the number of issued and outstanding shares of Common Stock at any time, except that the foregoing limit will not apply to awards outstanding as at July 3, 2023 to the extent such limit would be exceeded at July 3, 2023; and
- (e) the aggregate number of shares of Common Stock issuable to all insiders of the Company (as a group), pursuant to awards granted or issued in any twelve (12) month period under this Plan and all of the Company's other security-based compensation arrangements, shall not exceed ten percent (10%) of the number of issued and outstanding shares of Common Stock at the time of any grant of an award under this Plan to an insider.
- 4.4 Adjustments. If there shall occur any change with respect to the outstanding shares of Common Stock by reason of any recapitalization, reclassification, stock dividend, extraordinary cash dividend, stock split, reverse stock split or other distribution with respect to the shares of Common Stock, or any merger, reorganization, consolidation, combination, spin-off or other similar corporate change, or any other change affecting the Common Stock, the Committee shall, in the manner and to the extent equitable to the Participants and consistent with the terms of the Plan, cause an adjustment to be made in (i) the maximum number and kind of shares provided in Section 4.1 hereof, (ii) the number and kind of shares of Common Stock, units, or other rights subject to then outstanding Awards, the exercise or base price for each share or unit or other right subject to then outstanding Awards, and any other terms of an Award that are affected by the event. Notwithstanding the foregoing, any such adjustments shall, to the extent practicable, be made in a manner consistent with the requirements of Section 409A of the Code and, in the case of Incentive Stock Options, Section 424 of the Code.

5. Participation and Awards.

5.1 Designations of Participants. All Eligible Persons are eligible to be designated by the Committee to receive Awards and become Participants under the Plan. The Committee has the authority, in its discretion, to determine and designate from time to time those Eligible Persons who are to be granted Awards, the types of Awards to be granted and the number of shares of Common Stock or units subject to Awards granted under the Plan. In selecting Eligible Persons to be Participants and in determining the type and amount of Awards to be granted under the Plan, the Committee shall consider any and all factors that it deems relevant or appropriate.

5.2 Determination of Awards. The Committee shall determine the terms and conditions of all Awards granted to Participants in accordance with its authority under Section 3.2 hereof. An Award may consist of one type of right or benefit hereunder or of two or more such rights or benefits granted in tandem or in the alternative. In the case of any fractional share or unit resulting from the grant, vesting, payment or crediting of dividends or dividend equivalents under an Award, the Committee shall have the discretionary authority to (i) disregard such fractional share or unit, (ii) round such fractional share or unit to the nearest lower or higher whole share or unit, or (iii) convert such fractional share or unit into a right to receive a cash payment. To the extent deemed necessary by the Committee, an Award shall be evidenced by an Award Agreement as described in Section 15.1 hereof.

6. Stock Options.

- 6.1 Grant of Stock Options. A Stock Option may be granted to any Eligible Person selected by the Committee. Subject to the provisions of Section 6.6 hereof and Section 422 of the Code, each Stock Option shall be designated, in the discretion of the Committee, as an Incentive Stock Option or as a Nonqualified Stock Option. Investor relations service providers may not receive any awards other than Stock Options.
- 6.2 Exercise Price. The exercise price per share of a Stock Option shall not be less than of the Fair Market Value of the shares of Common Stock on the Date of Grant, provided, however, that the exercise price per share of an Incentive Stock Option granted to a 10% stockholder will not be less than 110% of the Fair Market Value of the shares of Common Stock on the Date of Grant.
- (a) Vesting of Stock Options. The Committee shall in its discretion prescribe the time or times at which, or the conditions upon which, a Stock Option or portion thereof shall become vested and/or exercisable, and may accelerate the vesting or exercisability of any Stock Option at any time. The requirements for vesting and exercisability of a Stock Option may be based on the continued Service of the Participant with the Company or its Subsidiaries for a specified time period (or periods) or on the attainment of specified performance goals established by the Committee in its discretion. In accordance with TSX policies, and subject to TSX approval to the contrary, and provided that the shares of Common Stock are then listed on the TSX, Stock Options granted to investor relations service providers must vest (and not otherwise be exercisable) in stages over a minimum of twelve (12) months, such that:
 - (i) no more than 1/4 of the Stock Options vest no sooner than three (3) months after the Stock Options were granted;
 - (ii) no more than another 1/4 of the Stock Options vest no sooner than six (6) months after the Stock Options were granted;
 - (iii) no more than another 1/4 of the Stock Options vest no sooner than nine (9) months after the Stock Options were granted; and
 - (iv) the remainder of the Stock Options vest no sooner than twelve (12) months after the Stock Options were granted.

- 6.3 Term of Stock Options. The Committee shall in its discretion prescribe in an Award Agreement the period during which a vested Stock Option may be exercised, provided that the maximum term of a Stock Option shall be ten (10) years from the Date of Grant. Except as provided in this Section 6 or as otherwise may be provided by the Committee, no Stock Option may be exercised at any time during the term thereof unless the Participant is then in the Service of the Company or one of its Subsidiaries.
- 6.4 Termination of Service. Subject to Section 6.6 hereof with respect to Incentive Stock Options, and except as may otherwise be provided in an Award Agreement, the Stock Option of any Participant whose Service with the Company or one of its Subsidiaries is terminated for any reason shall terminate on the earlier of (A) the date that the Stock Option expires in accordance with its terms or (B) the expiration of the applicable time period following termination of Service, in accordance with the following: (1) 12 months if Service ceased due to death or Disability, (2) 12 months if Service ceased due to Retirement, or (3) 90 days if Service ceased as a result of a termination by the Company without Cause or if Service ceased for any other reason; provided that, in the event of a termination for Cause such Participant's right to any further payments, vesting or exercisability with respect to any Award shall be forfeited in its entirety in accordance with Section 14. The Committee shall have authority to determine in each case whether an authorized leave of absence shall be deemed a termination of Service for purposes hereof, as well as the effect of a leave of absence on the vesting and exercisability of a Stock Option. Unless otherwise provided by the Committee, if an entity ceases to be a Subsidiary of, or to provide services to, the Company or otherwise ceases to be qualified under the Plan or if all or substantially all of the assets of a Subsidiary of the Company or an entity that provides services to the Company are conveyed (other than by encumbrance), such cessation or action, as the case may be, shall be deemed for purposes hereof to be a termination of the Service of all of the employees of the Subsidiary or other entity (unless at the time of the event they become employees of the Company).
- 6.5 Stock Option Exercise. Subject to such terms and conditions as shall be specified in an Award Agreement, a Stock Option may be exercised in whole or in part at any time during the term thereof by notice in the form required by the Company, together with payment of the aggregate exercise price and applicable withholding tax. Payment of the exercise price shall be made in the manner set forth in the Award Agreement, which unless otherwise provided by the Committee, may include:
 - (a) in cash or by cash equivalent acceptable to the Committee;
- (b) by payment in shares of Common Stock that have been held by the Participant for at least six months (or such greater or lesser period as the Committee may deem appropriate, for accounting purposes or otherwise) valued at the Fair Market Value of such shares on the date of exercise;
- (c) by a net exercise arrangement pursuant to which the Company will reduce the number of shares of Common Stock issued upon exercise by the largest whole number of shares of Common Stock with a Fair Market Value that does not exceed the aggregate exercise price ("Net Exercise");
- (d) to the extent permitted by law, through an open-market, broker-assisted sales transaction pursuant to which the Company is promptly delivered the amount of proceeds necessary to satisfy the exercise price;

- (e) by a combination of the methods described above; or
- (f) by such other method as may be approved by the Committee and set forth in the Award agreement,

provided that the Net Exercise alternative for payment may only be permitted in accordance with the procedures contemplated by the policies of the TSX so long as the shares of Common Stock are listed on the TSX.

6.6 Additional Rules for Incentive Stock Options.

- (a) Eligibility. An Incentive Stock Option may only be granted to an Eligible Person who is considered an employee for purposes of Treasury Regulation §1.421-7(h) with respect to the Company or any Subsidiary that qualifies as a "subsidiary corporation" with respect to the Company for purposes of Section 424(f) of the Code. In the event that an award is granted or issued to employees, consultants or management company employees, the Company and the Participant are responsible for ensuring and confirming that the optionee is a bona fide employee, consultant or management company employee, as the case may be, of the Company.
- (b) Annual Limits. Any portion of an Incentive Stock Option granted to a Participant as a result of which the aggregate Fair Market Value (determined as of the Date of Grant) of the stock with respect to which incentive stock options under Section 422 of the Code held by the Participant are exercisable for the first time in any calendar year under the Plan and any other stock option plans of the Company or any subsidiary or parent corporation, would exceed \$100,000, determined in accordance with Section 422(d) of the Code, shall become a Nonqualified Stock Option. This limitation shall be applied by taking stock options into account in the order in which granted.
- (c) Termination of Employment. An Award of an Incentive Stock Option may provide that such Stock Option may be exercised not later than 3 months following termination of employment of the Participant with the Company and all subsidiary corporations, or not later than one year following a permanent and total disability within the meaning of Section 22(e)(3) of the Code, as and to the extent determined by the Committee to comply with the requirements of Section 422 of the Code.
- (d) Other Terms and Conditions; Nontransferability. Any Stock Option that is not specifically designated as an Incentive Stock Option will under no circumstances be considered an Incentive Stock Option. Any Incentive Stock Option granted hereunder shall contain such additional terms and conditions, not inconsistent with the terms of the Plan, as are deemed necessary or desirable by the Committee, which terms, together with the terms of the Plan, shall be intended and interpreted to cause such Incentive Stock Option to qualify as an "incentive stock option" under Section 422 of the Code. An Award Agreement for an Incentive Stock Option may provide that such Stock Option shall be treated as a Nonqualified Stock Option to the extent that certain requirements applicable to "incentive stock options" under the Code shall not be satisfied. An Incentive Stock Option shall by its terms be nontransferable other than by will or by the laws of descent and distribution, and shall be exercisable during the lifetime of a Participant only by such Participant.

7. Stock Appreciation Rights.

- 7.1 Grant of Stock Appreciation Rights. A Stock Appreciation Right may be granted to any Eligible Person selected by the Committee. Stock Appreciation Rights may be granted on a basis that allows for the exercise of the right by the Participant or that provides for the automatic payment of the right upon a specified date or event. Stock Appreciation Rights shall be exercisable or payable at such time or times and upon such conditions as may be approved by the Committee, provided that the Committee may accelerate the exercisability or payment of a Stock Appreciation Right at any time.
- 7.2 Freestanding Stock Appreciation Rights. A Stock Appreciation Right may be granted without any related Stock Option and may be subject to such vesting and exercisability requirements as are specified by the Committee and described in an Award Agreement. Such vesting and exercisability requirements may be based on the continued Service of the Participant with the Company or its Subsidiaries for a specified time period (or periods) or on the attainment of specified performance goals established by the Committee in its discretion. A Stock Appreciation Right will be exercisable or payable at such time or times as determined by the Committee, provided that the maximum term of a Stock Appreciation Right shall be ten years from the Date of Grant. The base price of a Stock Appreciation Right granted without any related Stock Option shall be determined by the Committee in its sole discretion; provided, however, that the base price per share of any such freestanding Stock Appreciation Right shall not be less than 100% of the Fair Market Value of the shares of Common Stock on the Date of Grant.
- 7.3 Tandem Stock Option/Stock Appreciation Rights. A Stock Appreciation Right may be granted in tandem with a Stock Option, either at the time of grant or at any time thereafter during the term of the Stock Option. A tandem Stock Option/Stock Appreciation Right will entitle the holder to elect, as to all or any portion of the number of shares subject to such Stock Option/Stock Appreciation Right, to exercise either the Stock Option or the Stock Appreciation Right, resulting in the reduction of the corresponding number of shares subject to the right so exercised as well as the tandem right not so exercised. A Stock Appreciation Right granted in tandem with a Stock Option hereunder shall have a base price per share equal to the per share exercise price of the Stock Option, will become vested and exercisable at the same time or times that the related Stock Option becomes vested and exercisable, and will expire no later than the time at which the related Stock Option expires.
- 7.4 Payment of Stock Appreciation Rights. A Stock Appreciation Right will entitle the holder, upon exercise or other payment of the Stock Appreciation Right, as applicable, to receive an amount determined by multiplying: (i) the excess of the Fair Market Value of a share of Common Stock on the date of exercise or payment of the Stock Appreciation Right over the base price of such Stock Appreciation Right, by (ii) the number of shares as to which such Stock Appreciation Right is exercised or paid. Subject to the requirements of Section 409A of the Code, payment of the amount determined under the foregoing may be made, as approved by the Committee and set forth in the Award Agreement, in shares of Common Stock valued at their Fair Market Value on the date of exercise or payment, in cash, or in a combination of shares of Common Stock and cash, subject to applicable tax withholding requirements.

8. [intentionally deleted]

9. Restricted Stock Unit Awards.

- 9.1 Grant of Restricted Stock Unit Awards. A Restricted Stock Unit Award may be granted to any Eligible Person selected by the Committee. The value of each Restricted Stock Unit will be equal to the Fair Market Value of the Common Stock on the applicable date or time period of determination, as specified by the Committee. A Restricted Stock Unit Award shall be subject to such restrictions and conditions as the Committee shall determine. A Restricted Stock Unit Award may be granted together with a dividend equivalent right with respect to the shares of Common Stock subject to the Award, which may be accumulated and may be deemed reinvested in additional stock units, as determined by the Committee in its discretion.
- 9.2 Vesting Requirements. On the Date of Grant, the Committee shall in its discretion determine any vesting requirements with respect to a Restricted Stock Unit Award, which shall be set forth in the Award Agreement, provided that no Restricted Stock Unit Award may vest before the date that is one year following the date of grant or issue. Such vesting requirements may be based on the continued Service of the Participant with the Company or its Subsidiaries for a specified time period (or periods) or on the attainment of specified performance goals established by the Committee in its discretion. Notwithstanding the foregoing, the Committee may accelerate the vesting of a Restricted Stock Unit Award for a Participant who dies or who ceases to be an eligible Participant in connection with a Change of Control or similar transaction.
- 9.3 Payment of Restricted Stock Unit Awards. A Restricted Stock Unit Award shall become payable to a Participant at the time or times determined by the Committee and set forth in the Award Agreement, which may be upon or following the vesting of the Award. Payment of a Restricted Stock Unit Award may be made, at the discretion of the Committee, in cash or in shares of Common Stock, or in a combination thereof. Any cash payment of a Restricted Stock Unit Award shall be made based upon the Fair Market Value of the Common Stock, determined on such date or over such time period as determined by the Committee.
- 9.4 No Rights as Shareholder. The Participant shall not have any rights as a shareholder with respect to the shares subject to a Restricted Stock Unit Award until such time as shares of Common Stock are delivered to the Participant pursuant to the terms of the Award Agreement.

10. Performance Awards

- 10.1 Grant of Performance Awards. Performance Share and Performance Unit Awards (collectively, "Performance Awards") may be granted to any Eligible Person selected by the Committee. Performance Awards shall be subject to such restrictions and conditions as the Committee shall determine. Unless otherwise determined by the Committee at grant, a Performance Share Award shall not be granted with a dividend equivalent right with respect to the shares of Common Stock subject to the Award.
- 10.2 Vesting Requirements. On the Date of Grant, the Committee shall in its discretion determine any vesting requirements with respect to a Performance Award, which shall be set forth in the Award Agreement, provided that the Committee provided that no Performance Award may vest before the date that is one year following the date of grant or issue. Notwithstanding the foregoing, the Committee may accelerate the vesting of a Performance Award for a Participant who dies or who ceases to be an eligible Participant in connection with a Change of Control or similar transaction. Vesting requirements may be based on the continued Service of the Participant with the Company or its Subsidiaries for a specified time period (or periods), as well as on the attainment of specified performance goals established by the Committee in its discretion. The Committee may provide that if performance relative to the performance goals exceeds targeted levels, then the number of Performance Awards earned shall be a multiple (e.g., 150%) of those that would be earned for target performance.

- 10.3 Payment of Performance Awards. A Performance Award shall become payable to a Participant at the time or times determined by the Committee and set forth in the Award Agreement, which may be upon or following the vesting of the Award. Payment of a Performance Award may be made, at the discretion of the Committee, in cash or in shares of Common Stock, or in a combination thereof.
- 10.4 No Rights as Shareholder. The Participant shall not have any rights as a shareholder with respect to the shares subject to a Performance Share Award until such time as shares of Common Stock are delivered to the Participant pursuant to the terms of the Award Agreement.

11. [intentionally deleted].

12. Formula Awards to Directors.

- 12.1 General. Non-Employee Directors will be entitled to receive all types of Awards (except Incentive Stock Options) under the Plan, including Awards not covered under this Section 12. All grants of Awards to Non-Employee Directors pursuant to this Section 12 will be automatic and nondiscretionary.
- 12.2 No Discretion. The Committee shall have no authority, discretion or power to select the Non-Employee Directors who will receive any Award under this Section 12, determine the number of shares of Common Stock to be covered by any such Awards or the time at which such Awards are to be granted, establish the duration of the Awards or alter any other terms or conditions specified in the Plan, except in the sense of administering the Plan pursuant to the provisions of the Plan.
- 12.3 Grant of Stock Options. Each Non-Employee Director shall be automatically granted a Nonqualified Stock Option to purchase 25,000 shares of Common Stock (subject to adjustment pursuant to Section 4.4 hereof), on May 1 of each year following the Effective Date, if as of such Date of Grant, such Non-Employee Director will have served on the Board for at least the preceding three (3) months, which Stock Option shall become exercisable one year from the applicable Date of Grant. Stock Options shall expire, to the extent not exercised, five years after the date on which they were granted.
- 12.4 Price of Stock Options. The exercise price per share of Common Stock for any Stock Option granted pursuant to this Section 12 shall be the greater of (a) 100% of the Fair Market Value of a share of Common Stock on the date of grant of the Stock Option and (b) the minimum price permitted by the rules and policies of any stock exchange on which the shares of Common Stock are then listed.
- 12.5 Other Terms. Except for the limitations set forth in this Section 12 and elsewhere in the Plan, the terms and provisions of the Stock Options granted pursuant to this Section 12 shall be as determined from time to time by the Committee, and Stock Options issued pursuant to this Section 12 may contain terms and provisions different from other Stock Options granted to the same or other Stock Option recipients. Stock Options shall be evidenced by an Award Agreement containing such terms and provisions as the Committee may determine, subject to the provisions of this Section 12 and the Plan.

13. Change in Control.

13.1 Effect of Change in Control The Committee may, in its discretion determine, and provide in the applicable Award Agreement, that vesting or other terms of an Award will be accelerated or otherwise affected by a Change in Control, either alone or together with a termination of employment without Cause or by the Participant for Good Reason or by the Company without Cause, or otherwise, subject to shareholder approval if required by applicable stock exchange listing rules.

14. Forfeiture Events.

- 14.1 General. The Committee may specify in an Award Agreement at the time of the Award that the Participant's rights, payments and benefits with respect to an Award shall be subject to reduction, cancellation, forfeiture or recoupment upon the occurrence of certain specified events, in addition to any otherwise applicable vesting or performance conditions of an Award. Such events may include, but shall not be limited to, violation of material Company policies, breach of noncompetition, confidentiality or other restrictive covenants that may apply to the Participant, or other conduct by the Participant that is detrimental to the business or reputation of the Company or its Subsidiaries.
- 14.2 Termination for Cause. If a Participant's employment with the Company or any Subsidiary shall be terminated for Cause, such Participant's right to any further payments, vesting or exercisability with respect to any Award shall terminate in its entirety.

15. General Provisions.

- 15.1 Award Agreement. To the extent deemed necessary by the Committee, an Award under the Plan shall be evidenced by an Award Agreement in a written or electronic form. The Award Agreement shall be subject to and incorporate, by reference or otherwise, all of the applicable terms and conditions of the Plan, and may also set forth other terms and conditions applicable to the Award as determined by the Committee consistent with the limitations of the Plan. The grant of an Award under the Plan shall not confer any rights upon the Participant holding such Award other than such terms, and subject to such conditions, as are specified in the Plan as being applicable to such type of Award (or to all Awards) or as are expressly set forth in the Award Agreement. The Committee need not require the execution of an Award Agreement by a Participant, in which case, acceptance of the Award by the Participant shall constitute agreement by the Participant to the terms, conditions, restrictions and limitations set forth in the Plan and the Award Agreement as well as the administrative guidelines of the Company in effect from time to time.
- 15.2 No Assignment or Transfer; Beneficiaries. Unless otherwise expressly provided in (or pursuant to) this Plan and permitted by applicable law and the rules and policies of any stock exchange on which the shares of Common Stock are then listed (including the TSX), awards under the Plan are non-assignable and non-transferable by the Participant, except by will or by the laws of descent and distribution, and shall not be subject in any manner to assignment, alienation, pledge, encumbrance or charge. Notwithstanding the foregoing, the Committee may provide in an Award Agreement that the Participant shall have the right to designate a beneficiary or beneficiaries who shall be entitled to any rights, payments or other benefits specified under an Award following the Participant's death. During the lifetime of a Participant, an Award shall be exercised only by such Participant or such Participant's guardian or legal representative. In the event of a Participant's death, an Award may to the extent permitted by the Award Agreement be exercised by the Participant's beneficiary as designated by the Participant in the manner prescribed by the Committee or, in the absence of an authorized beneficiary designation, by the legatee of such Award under the Participant's will or by the Participant's estate in accordance with the Participant's will or the laws of descent and distribution, in each case in the same manner and to the same extent that such Award was exercisable by the Participant on the date of the Participant's death.

- 15.3 Deferrals of Payment. The Committee may in its discretion permit a Participant to defer the receipt of payment of cash or delivery of shares of Common Stock that would otherwise be due to the Participant by virtue of the exercise of a right or the satisfaction of vesting or other conditions with respect to an Award. If any such deferral is to be permitted by the Committee, the Committee shall establish rules and procedures relating to such deferral in a manner intended to comply with the requirements of Section 409A of the Code, including, without limitation, the time when an election to defer may be made, the time period of the deferral and the events that would result in payment of the deferred amount, the interest or other earnings attributable to the deferral and the method of funding, if any, attributable to the deferred amount.
- 15.4 Rights as Shareholder. A Participant shall have no rights as a holder of shares of Common Stock with respect to any unissued securities covered by an Award until the date the Participant becomes the holder of record of such securities. Except as provided in Section 4.4 hereof, no adjustment or other provision shall be made for dividends or other shareholder rights, except to the extent that the Award Agreement provides for dividend payments or dividend equivalent rights.
- 15.5 Employment or Service. Nothing in the Plan, in the grant of any Award or in any Award Agreement shall confer upon any Eligible Person any right to continue in the Service of the Company or any of its Subsidiaries, or interfere in any way with the right of the Company or any of its Subsidiaries to terminate the Participant's employment or other service relationship for any reason at any time.
- 15.6 Securities Laws. No shares of Common Stock will be issued or transferred pursuant to an Award unless and until all then applicable requirements imposed by Federal and state securities and other laws, rules and regulations and by any regulatory agencies having jurisdiction, and by any exchanges or markets upon which the shares of Common Stock may be listed, have been fully met. As a condition precedent to the issuance of shares pursuant to the grant or exercise of an Award, the Company may require the Participant to take any reasonable action to meet such requirements. The Committee may impose such conditions on any shares of Common Stock issuable under the Plan as it may deem advisable, including, without limitation, restrictions in order to ensure compliance with the Securities Act of 1933, as amended, the requirements of any exchange or market upon which such shares of the same class are then listed, and any blue sky or other securities laws applicable to such shares. The Committee may also require the Participant to represent and warrant at the time of issuance or transfer that the shares of Common Stock are being acquired only for investment purposes and without any current intention to sell or distribute such shares.

- 15.7 Tax Withholding. The Participant shall be responsible for payment of any taxes or similar charges required by law to be withheld from an Award or an amount paid in satisfaction of an Award, which shall be paid by the Participant on or prior to the payment or other event that results in taxable income in respect of an Award. The Award Agreement may specify the manner in which the withholding obligation shall be satisfied with respect to the particular type of Award, which may include procedures to permit or require a Participant to satisfy such obligation in whole or in part (but only up to the statutory minimum) by having the Company withhold shares of Common Stock from the shares to which the Participant is entitled. The number of shares to be withheld shall have a Fair Market Value as of the date that the amount of tax to be withheld is determined as nearly equal as possible to (but not exceeding) the amount of such obligations being satisfied. Notwithstanding the foregoing, the Company, in its sole discretion, may withhold all such required taxes from any amount otherwise payable to a Participant. Notwithstanding anything contained in the Plan or any Award Agreement to the contrary, a Participant's satisfaction of any tax-withholding requirements will be a condition precedent to the Company's obligation to issue Common Stock or make payments to that Participant as may otherwise be provided and to the termination of any restrictions on transfer related to the circumstance or event that results in the tax-withholding requirement.
- 15.8 Unfunded Plan. The adoption of the Plan and any reservation of shares of Common Stock or cash amounts by the Company to discharge its obligations hereunder shall not be deemed to create a trust or other funded arrangement. Except upon the issuance of Common Stock pursuant to an Award, any rights of a Participant under the Plan shall be those of a general unsecured creditor of the Company, and neither a Participant nor the Participant's permitted transferees or estate shall have any other interest in any assets of the Company by virtue of the Plan. Notwithstanding the foregoing, the Company shall have the right to implement or set aside funds in a grantor trust, subject to the claims of the Company's creditors or otherwise, to discharge its obligations under the Plan.
- 15.9 Other Compensation and Benefit Plans. The adoption of the Plan shall not affect any other share incentive or other compensation plans in effect for the Company or any Subsidiary, nor shall the Plan preclude the Company from establishing any other forms of share incentive or other compensation or benefit program for employees of the Company or any Subsidiary. The amount of any compensation deemed to be received by a Participant pursuant to an Award shall not constitute includable compensation for purposes of determining the amount of benefits to which a Participant is entitled under any other compensation or benefit plan or program of the Company or a Subsidiary, including, without limitation, under any pension or severance benefits plan, except to the extent specifically provided by the terms of any such plan.
- 15.10 Plan Binding on Transferees. The Plan shall be binding upon the Company, its transferees and assigns, each Participant, and each Participant's executor, administrator and permitted transferees and beneficiaries.
- 15.11 Severability. If any provision of the Plan or any Award Agreement shall be determined to be illegal or unenforceable by any court of law in any jurisdiction, the remaining provisions hereof and thereof shall be severable and enforceable in accordance with their terms, and all provisions shall remain enforceable in any other jurisdiction.
- 15.12 Foreign Jurisdictions. The Committee may adopt, amend and terminate such arrangements and grant such Awards, not inconsistent with the intent of the Plan, as it may deem necessary or desirable to comply with any tax, securities, regulatory or other laws of jurisdictions outside the United States of America with respect to Awards that may be subject to such laws. The terms and conditions of such Awards may vary from the terms and conditions that would otherwise be required by the Plan solely to the extent the Committee deems necessary for such purpose. Moreover, the Board may approve such supplements to or amendments, restatements or alternative versions of the Plan, not inconsistent with the intent of the Plan, as it may consider necessary or appropriate for such purposes, without thereby affecting the terms of the Plan as in effect for any other purpose.

- 15.13 Substitute Awards in Corporate Transactions. Nothing contained in the Plan shall be construed to limit the right of the Committee to grant Awards under the Plan in connection with the acquisition, whether by purchase, merger, consolidation or other corporate transaction, of the business or assets of any corporation or other entity. Without limiting the foregoing, the Committee may grant Awards under the Plan to an employee or director of another corporation who becomes an Eligible Person by reason of any such corporate transaction in substitution for awards previously granted by such corporation or entity to such. The terms and conditions of the substitute Awards may vary from the terms and conditions that would otherwise be required by the Plan solely to the extent the Committee deems necessary for such purpose.
- 15.14 No Fiduciary Relationship. Nothing in the Plan and no action taken pursuant to the Plan, will create a fiduciary relationship between the Company, its Directors or officers or the Committee, on the one hand, and the Participant or any other person or entity, on the other.
- 15.15 Compliance with Section 409A. Unless otherwise expressly provided for in an Award Agreement, the Plan and Award Agreements will be interpreted to the greatest extent possible in a manner that makes the Plan and the Awards granted hereunder exempt from Section 409A of the Code, and, to the extent not so exempt, in compliance with Section 409A of the Code. If the Board determines that any Award granted hereunder is not exempt from and is therefore subject to Section 409A of the Code, the Award Agreement evidencing such Award will incorporate the terms and conditions necessary to avoid the consequences specified in Section 409A(a)(1) of the Code, and to the extent an Award Agreement is silent on terms necessary for compliance, such terms are hereby incorporated by reference into the Award Agreement.
- 15.16 Governing Law. The Plan and all rights hereunder shall be subject to and interpreted in accordance with the laws of the State of Delaware, without reference to the principles of conflicts of laws, and to applicable Federal securities laws.
- 15.17 Employment and Other Agreements. Any provision in a written employment, severance or other agreement providing accelerated vesting or other protective provisions applicable to types of Awards granted under the Plan will apply to Awards under the Plan unless the Award Agreement relating to a particular Award provides otherwise.
- 15.18 *Notices*. All notices under the Plan must be in writing or delivered electronically, if to the Company, at its principal office, addressed to the attention of the Chief Executive Officer; and if to the Participant, at the address appearing in the Company's records.
 - 15.19 Captions. The use of captions in this Plan is for convenience. The captions are not intended to provide substantive rights.

16. Effective Date; Amendment and Termination.

- 16.1 Effective Date. The Plan shall become effective following its adoption by the Board (the "Effective Date"), subject to its approval by the Company's shareholders within twelve (12) months after such adoption by the Board to the extent then required under Section 422 or 424 of the Code or any other applicable law, or deemed necessary or advisable by the Board.
- 16.2 Amendment. The Board may at any time and from time to time and in any respect, amend or modify the Plan and any Award granted under the Plan. The Board may seek the approval of any amendment or modification by the Company's shareholders to the extent it deems necessary or advisable in its discretion for purposes of compliance with Section 162(m) or Section 422 of the Code, the listing requirements of the applicable exchange or securities market or for any other purpose, provided, however, that the Company shall be required to seek shareholder approval of amendments to Awards if required by applicable stock exchange listing rules. Except as provided elsewhere herein, no amendment or modification of the Plan or any Award shall adversely affect any Award theretofore granted without the consent of the Participant or the permitted transferee of the Award. Notwithstanding the foregoing or any other provision of this Plan, disinterested stockholder approval must be obtained for any reduction in the exercise price of a Stock Option, or the extension of the term of a Stock Option, if the Participant is an insider of the Company at the time of the proposed amendment.
- 16.3 Termination. The Plan shall terminate at the close of business on the day before the $10^{\rm h}$ anniversary of the Effective Date. The Board may, in its discretion and at any earlier date, terminate the Plan. Notwithstanding the foregoing, no termination of the Plan shall adversely affect any Award theretofore granted without the consent of the Participant or the permitted transferee of the Award.

BIRCHTECH CORP. (formerly Midwest Energy Emissions Corp.)

AMENDED AND RESTATED 2017 EQUITY INCENTIVE PLAN Adopted July 3, 2023; Amended October 29, 2024

1. PURPOSE OF PLAN; GENERAL

- 1.1 The purpose of this 2017 Equity Incentive Plan (this "Plan") of Birchtech Corp., a Delaware corporation (the "Corporation"), is to promote the success of the Corporation and to increase stockholder value by providing an additional means through the grant of awards to attract, motivate, retain and reward selected employees and other eligible persons.
- 1.2 For the purposes of the Plan, the terms "employee", "consultant", "consultant company", "management company employee", "investor relations activities", "insiders" and "investor relations service providers" shall have the respective meanings ascribed thereto in the policies of the Toronto Stock Exchange ("TSX") or any other applicable stock exchange.

2. ELIGIBILITY

- 2.1 The Administrator (as such term is defined in Section 3.1) may grant awards under this Plan only to those persons that the Administrator determines to be Eligible Persons. An "Eligible Person" is any person who is either: (a) an officer (whether or not a director) or employee of the Corporation or one of its Subsidiaries; (b) a director of the Corporation or one of its Subsidiaries; or (c) a consultant to the Corporation or one of its Subsidiaries and who is selected to participate in this Plan by the Administrator. An Eligible Person who has been granted an award (a "participant") may, if otherwise eligible, be granted additional awards if the Administrator shall so determine. As used herein, "Subsidiary" means any corporation or other entity a majority of whose outstanding voting stock or voting power is beneficially owned directly or indirectly by the Corporation; and "Board" means the Board of Directors of the Corporation.
- 2.2 In the event that an award is granted or issued to employees, consultants or management company employees, the Corporation and the participant are responsible for ensuring and confirming that the optionee is a *bona fide* employee, consultant or management company employee, as the case may be, of the Corporation.

3. PLAN ADMINISTRATION

3.1 The Administrator. This Plan shall be administered by and all awards under this Plan shall be authorized by the Administrator. The "Administrator" means the Board or one or more committees appointed by the Board or another committee (within its delegated authority) to administer all or certain aspects of this Plan. Any such committee shall be comprised solely of one or more directors or such number of directors as may be required under applicable law. A committee may delegate some or all of its authority to another committee so constituted. The Board or a committee comprised solely of directors may also delegate, to the extent permitted by Section 157 of the Delaware General Corporation Law and any other applicable law, to one or more officers of the Corporation, its powers under this Plan (a) to determine Eligible Persons who will receive grants of awards under this Plan, and (b) to determine the number of shares subject to, and the other terms and conditions of, such awards. The Board may delegate different levels of authority to different committees with administrative and grant authority under this Plan. Unless otherwise provided in the bylaws of the Corporation or the applicable charter of any Administrator: (a) a majority of the members of the acting Administrator shall constitute a quorum, and (b) the affirmative vote of a majority of the members present assuming the presence of a quorum or the unanimous written consent of the members of the Administrator shall constitute due authorization of an action by the acting Administrator.

With respect to awards intended to satisfy the requirements for performance-based compensation under Section 162(m) of the Internal Revenue Code of 1986, as amended (the "Code"), this Plan shall be administered by a committee consisting solely of two or more outside directors (as this requirement is applied under Section 162(m) of the Code); provided, however, that the failure to satisfy such requirement shall not affect the validity of the action of any committee otherwise duly authorized and acting in the matter. Award grants, and transactions in or involving awards, intended to be exempt under Rule 16b-3 under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), must be duly and timely authorized by the Board or a committee consisting solely of two or more non-employee directors or otherwise comply with such other conditions of the Rule to satisfy such exemption provisions (as this requirement is applied under Rule 16b-3 promulgated under the Exchange Act). To the extent required by any applicable stock exchange, this Plan shall be administered by a committee composed entirely of independent directors (within the meaning of the applicable stock exchange). Awards granted to non-employee directors shall not be subject to the discretion of any officer or employee of the Corporation and shall be administered exclusively by a committee consisting solely of independent directors or approved by the Board as a whole.

- **3.2 Powers of the Administrator.** Subject to the express provisions of this Plan, the Administrator is authorized and empowered to do all things necessary or desirable in connection with the authorization of awards and the administration of this Plan (in the case of a committee or delegation to one or more officers, within the authority delegated to that committee or person(s)), including, without limitation, the authority to:
- (a) determine eligibility and, from among those persons determined to be eligible, the particular Eligible Persons who will receive awards under this Plan;
- (b) grant awards to Eligible Persons, determine the price at which securities will be offered or awarded and the number of securities to be offered or awarded to any of such persons, determine the other specific terms and conditions of such awards consistent with the express limits of this Plan, establish the installments (if any) in which such awards shall become exercisable or shall vest (which may include, without limitation, performance and/or time-based schedules), or determine that no delayed exercisability or vesting is required, establish any applicable performance targets, and establish the events of termination or reversion of such awards;
 - (c) approve the forms of award agreements (which need not be identical either as to type of award or among participants);
- (d) construe and interpret this Plan and any agreements defining the rights and obligations of the Corporation, its Subsidiaries, and participants under this Plan, further define the terms used in this Plan, and prescribe, amend and rescind rules and regulations relating to the administration of this Plan or the awards granted under this Plan;

- (e) cancel, modify, or waive the Corporation's rights with respect to, or modify, discontinue, suspend, or terminate any or all outstanding awards, subject to any required consent under Section 8.6.5;
- (f) accelerate or extend the vesting or exercisability or extend the term of any or all such outstanding awards (in the case of options or stock appreciation rights, within the maximum ten (10) year term of such awards) in such circumstances as the Administrator may deem appropriate (including, without limitation, in connection with a termination of employment or services or other events of a personal nature) subject to any required consent under Section 8.6.5;
- (g) adjust the number of shares of Common Stock subject to any award, adjust the price of any or all outstanding awards or otherwise change previously imposed terms and conditions, in such circumstances as the Administrator may deem appropriate, in each case subject to compliance with applicable stock exchange requirements, Sections 4 and 8.6 and the applicable requirements of Code Section 162(m) and treasury regulations thereunder with respect to awards that are intended to satisfy the requirements for performance-based compensation under Section 162(m), and provided that in no case (except due to an adjustment contemplated by Section 7 or any repricing that may be approved by stockholders) shall such an adjustment constitute a repricing (by amendment, cancellation and regrant, exchange or other means) of the per share exercise or base price of any stock option or stock appreciation right or other award granted under this Plan, and further provided that any adjustment or change in terms made pursuant to this Section 3.2(g) shall be made in a manner that, in the good faith determination of the Administrator will not likely result in the imposition of additional taxes or interest under Section 409A of the Code;
- (h) determine the date of grant of an award, which may be a designated date after but not before the date of the Administrator's action (unless otherwise designated by the Administrator, the date of grant of an award shall be the date upon which the Administrator took the action granting an award);
- (i) determine whether, and the extent to which, adjustments are required pursuant to Section 7 hereof and authorize the termination, conversion, substitution, acceleration or succession of awards upon the occurrence of an event of the type described in Section 7;
 - (j) acquire or settle (subject to Sections 7 and 8.6) rights under awards in cash, stock of equivalent value or other consideration; and
- (k) determine the Fair Market Value (as defined in Section 5.6) of the Common Stock or awards under this Plan from time to time and/or the manner in which such value will be determined.
- 3.3 Binding Determinations. Any action taken by, or inaction of, the Corporation, any Subsidiary, or the Administrator relating or pursuant to this Plan and within its authority hereunder or under applicable law shall be within the absolute discretion of that entity or body and shall be conclusive and binding upon all persons. Neither the Board, the Administrator, nor any Board committee, nor any member thereof or person acting at the direction thereof, shall be liable for any act, omission, interpretation, construction or determination made in good faith in connection with this Plan (or any award made under this Plan), and all such persons shall be entitled to indemnification and reimbursement by the Corporation in respect of any claim, loss, damage or expense (including, without limitation, legal fees) arising or resulting therefrom to the fullest extent permitted by law and/or under any directors and officers liability insurance coverage that may be in effect from time to time.

- **3.4 Reliance on Experts.** In making any determination or in taking or not taking any action under this Plan, the Administrator may obtain and may rely upon the advice of experts, including professional advisors to the Corporation. The Administrator shall not be liable for any such action or determination taken or made or omitted in good faith based upon such advice.
- **3.5 Delegation of Non-Discretionary Functions.** In addition to the ability to delegate certain grant authority to officers of the Corporation as set forth in Section 3.1, the Administrator may also delegate ministerial, non-discretionary functions to individuals who are officers or employees of the Corporation or any of its Subsidiaries or to third parties.

4. SHARES OF COMMON STOCK SUBJECT TO THE PLAN; SHARE LIMIT

- **4.1** Shares Available. Subject to the provisions of Section 7.1, the capital stock available for issuance under this Plan shall be shares of the Corporation's authorized but unissued Common Stock. For purposes of this Plan, "Common Stock" shall mean the common stock of the Corporation and such other securities or property as may become the subject of awards under this Plan, or may become subject to such awards, pursuant to an adjustment made under Section 7.1.
- 4.2 Share Limit. Prior to July 3, 2023, an aggregate of 1,502,427 shares of Common Stock have been issued under the Plan. The maximum number of shares of Common Stock that may be issued under the Plan after July 3, 2023 shall be 14,078,459 (the "Share Limit"). To the extent any award (or portion thereof) outstanding under the Corporation's 2014 Equity Incentive Plan expire, terminate or are cancelled, surrendered or forfeited for any reason on or after July 3, 2023, the shares of Common Stock subject to such award (or portion thereof) shall be added to and increase the Share Limit, to a maximum of 4,775,000 additional shares of Common Stock. The foregoing Share Limit is subject to adjustment as contemplated by Section 4.4, Section 7.1, and Section 8.10.
- **4.3** *Limitations on Awards.* Except where permitted by the policies of the TSX (including, without limitation, upon obtaining any disinterested stockholder approval required therefor) and so long as the shares of Common Stock are listed on the TSX:
- **4.3.1** the aggregate number of shares of Common Stock issuable to any one participant (and any companies wholly owned by such participant), pursuant to awards granted or issued in any twelve (12) month period under this Plan and all of the Corporation's other security-based compensation arrangements, shall not exceed five percent (5%) of the number of issued and outstanding shares of Common Stock at the date the award is granted;
- **4.3.2** the aggregate number of shares of Common Stock issuable to any one consultant, pursuant to awards granted or issued in any twelve (12) month period under this Plan and all of the Corporation's other security-based compensation arrangements, shall not exceed two percent (2%) of the number of issued and outstanding shares of Common Stock at the time the award is granted to the said consultant under this Plan;
- **4.3.3** the aggregate number of shares of Common Stock issuable to all investor relations service providers, pursuant to awards granted or issued in any twelve (12) month period under this Plan and all of the Corporation's other security-based compensation arrangements, shall not exceed two percent (2%) of the number of issued and outstanding shares of Common Stock at the time of any grant of an award under this Plan to an investor relations service provider;

- 4.3.4 the aggregate number of shares of Common Stock issuable to insiders of the Corporation (as a group), pursuant to awards granted or issued under this Plan and all of the Corporation's other security-based compensation arrangements, shall not exceed ten percent (10%) of the number of issued and outstanding shares of Common Stock at any time, except that the foregoing limit will not apply to awards outstanding as at July 3, 2023 to the extent such limit would be exceeded at July 3, 2023; and
- **4.3.5** the aggregate number of shares of Common Stock issued to all insiders of the Corporation (as a group), pursuant to awards granted or issued in any twelve (12) month period under this Plan and all of the Corporation's other security-based compensation arrangements, shall not exceed ten percent (10%) of the number of issued and outstanding shares of Common Stock at the time of any grant of an award under this Plan to an insider.
- 4.4 Awards Settled in Cash, Reissue of Awards and Shares. The Administrator may adopt reasonable counting procedures to ensure appropriate counting, avoid double counting (as, for example, in the case of tandem or substitute awards) and make adjustments in accordance with this Section 4.4. Shares shall be counted against those reserved to the extent such shares have been delivered and are no longer subject to a substantial risk of forfeiture. Accordingly, to the extent that an award under the Plan, in whole or in part, is canceled, expired, forfeited, settled in cash, settled by delivery of fewer shares than the number of shares underlying the award, or otherwise terminated without delivery of shares to the participant, the shares retained by or returned to the Corporation will not be deemed to have been delivered under the Plan and will be deemed to remain or to become available under this Plan; and (ii) shares that are withheld from such an award or separately surrendered by the participant in payment of taxes relating to such an award shall be deemed to constitute shares not delivered and will be deemed to remain or to become available under the Plan. The foregoing adjustments to the Share Limit of this Plan are subject to any applicable limitations under Section 162(m) of the Code with respect to awards intended as performance-based compensation thereunder.
- **4.5** Reservation of Shares; No Fractional Shares. The Corporation shall at all times reserve a number of shares of Common Stock sufficient to cover the Corporation's obligations and contingent obligations to deliver shares with respect to awards then outstanding under this Plan (exclusive of any dividend equivalent obligations to the extent the Corporation has the right to settle such rights in cash). No fractional shares shall be delivered under this Plan. The Administrator may pay cash in lieu of any fractional shares in settlements of awards under this Plan.

5. AWARDS

- **5.1** Type and Form of Awards. Subject to the provisions of this Plan, the Administrator shall determine the type or types of award(s) to be made to each selected Eligible Person. Awards may be granted singly, in combination or in tandem. Awards also may be made in combination or in tandem with, in replacement of, as alternatives to, or as the payment form for grants or rights under any other employee or compensation plan of the Corporation or one of its Subsidiaries. The types of awards that may be granted under this Plan are:
- **5.1.1** Stock Options. A stock option is the grant of a right to purchase a specified number of shares of Common Stock during a specified period as determined by the Administrator. An option may be intended as an incentive stock option within the meaning of Section 422 of the Code (an "ISO") or a nonqualified stock option (an option not intended to be an ISO). The award agreement for an option will indicate if the option is intended as an ISO; otherwise it will be deemed to be a nonqualified stock option. The maximum term of each option (ISO or nonqualified) shall be ten (10) years. The per share exercise price for each option shall be not less than the greater of (a) 100% of the Fair Market Value of a share of Common Stock on the date of grant of the option and (b) the minimum price permitted by the rules and policies of any stock exchange on which the shares of Common Stock are then listed. When an option is exercised, the exercise price for the shares to be purchased shall be paid in full in cash or such other method permitted by the Administrator consistent with Section 5.5. Investor relations service providers may not receive any awards other than options.

5.1.2 Additional Rules Applicable to ISOs. To the extent that the aggregate Fair Market Value (determined at the time of grant of the applicable option) of stock with respect to which ISOs first become exercisable by a participant in any calendar year exceeds \$100,000, taking into account both Common Stock subject to ISOs under this Plan and stock subject to ISOs under all other plans of the Corporation or one of its Subsidiaries (or any parent or predecessor corporation to the extent required by and within the meaning of Section 422 of the Code and the regulations promulgated thereunder), such options shall be treated as nonqualified stock options. In reducing the number of options treated as ISOs to meet the \$100,000 limit, the administrator may, in the manner and to the extent a reduction of simultaneously granted options is necessary to meet the \$100,000 limit, the Administrator may, in the manner and to the extent permitted by law, designate which shares of Common Stock are to be treated as shares acquired pursuant to the exercise of an ISO. ISOs may only be granted to employees of the Corporation or one of its subsidiaries (for this purpose, the term "subsidiary" is used as defined in Section 424(f) of the Code, which generally requires an unbroken chain of ownership of at least 50% of the total combined voting power of all classes of stock of each subsidiary in the chain beginning with the Corporation and ending with the subsidiary in question). There shall be imposed in any award agreement relating to ISOs such other terms and conditions as from time to time are required in order that the option be an "incentive stock option" as that term is defined in Section 422 of the Code. No ISO may be granted to any person who, at the time the option is granted, owns (or is deemed to own under Section 424(d) of the Code) shares of outstanding Common Stock possessing more than 10% of the total combined voting power of all classes of stock of the Corporation, unless the exercise price of such option is at least 110%

5.1.3 Stock Appreciation Rights. A stock appreciation right or "SAR" is a right to receive a payment, in cash and/or Common Stock, equal to the number of shares of Common Stock being exercised multiplied by the excess of (i) the Fair Market Value of a share of Common Stock on the date the SAR is exercised, over (ii) the Fair Market Value of a share of Common Stock on the date the SAR was granted as specified in the applicable award agreement (the "base price"). The maximum term of a SAR shall be ten (10) years.

5.1.4 [intentionally deleted]

5.1.5 Restricted Share Units.

(a) Grant of Restricted Share Units. A restricted share unit, or "RSU", represents the right to receive from the Corporation on the respective scheduled vesting or payment date for such RSU, one share of Common Stock. An award of RSUs may be subject to the attainment of specified performance goals or targets, forfeitability provisions and such other terms and conditions as the Administrator may determine, subject to the provisions of this Plan. At the time an award of RSUs is made, the Administrator shall establish a period of time during which the restricted share units shall vest and the timing for settlement of the RSU.

- (b) Dividend Equivalent Accounts. Subject to the terms and conditions of the Plan and the applicable award agreement, as well as any procedures established by the Administrator, prior to the expiration of the applicable vesting period of an RSU, the Administrator may determine to pay dividend equivalent rights with respect to RSUs, in which case, the Corporation shall establish an account for the participant and reflect in that account any securities, cash or other property comprising any dividend or property distribution with respect to the shares of Common Stock underlying each RSU. Each amount or other property credited to any such account shall be subject to the same vesting conditions as the RSU to which it relates. The participant shall have the right to be paid the amounts or other property credited to such account upon vesting of the subject RSU.
- (c) Rights as a Stockholder. Subject to the restrictions imposed under the terms and conditions of this Plan and the applicable award agreement, each participant receiving RSUs shall have no rights as a stockholder with respect to such RSUs until such time as shares of Common Stock are issued to the participant. No shares of Common Stock shall be issued at the time a RSU is granted, and the Corporation will not be required to set aside a fund for the payment of any such award. Except as otherwise provided in the applicable award agreement, shares of Common Stock issuable under an RSU shall be treated as issued on the first date that the holder of the RSU is no longer subject to a substantial risk of forfeiture as determined for purposes of Section 409A of the Code, and the holder shall be the owner of such shares of Common Stock on such date. An award agreement may provide that issuance of shares of Common Stock under an RSU may be deferred beyond the first date that the RSU is no longer subject to a substantial risk of forfeiture, provided that such deferral is structured in a manner that is intended to comply with the requirements of Section 409A of the Code.
- **5.1.6** Cash Awards. The Administrator may, from time to time, subject to the provisions of the Plan and such other terms and conditions as it may determine, grant cash bonuses (including without limitation, discretionary awards, awards based on objective or subjective performance criteria, awards subject to other vesting criteria or awards granted consistent with Section 5.2 below). Cash awards shall be awarded in such amount and at such times during the term of the Plan as the Administrator shall determine.
- 5.1.5 through 5.1.6 above may be, and options and SARs granted with an exercise or base price not less than the Fair Market Value of a share of Common Stock at the date of grant ("Qualifying Options" and "Qualifying SARs," respectively) typically will be, granted as awards intended to satisfy the requirements for "performance-based compensation" within the meaning of Section 162(m) of the Code ("Performance-Based Awards"). The grant, vesting, exercisability or payment of Performance-Based Awards may depend (or, in the case of Qualifying Options or Qualifying SARs, may also depend) on the degree of achievement of one or more performance goals relative to a pre-established targeted level or levels using the Business Criteria provided for below for the Corporation on a consolidated basis or for one or more of the Corporation's Subsidiaries, segments, divisions or business units, or any combination of the foregoing. Such criteria may be evaluated on an absolute basis or relative to prior periods, industry peers or stock market indices. Any Qualifying Option or Qualifying SAR shall be subject to the requirements of Section 5.2.1 and 5.2.3 in order for such award to satisfy the requirements for "performance-based compensation" under Section 162(m) of the Code. Any other Performance-Based Award shall be subject to all of the following provisions of this Section 5.2.

- **5.2.1** Class; Administrator. The eligible class of persons for Performance-Based Awards under this Section 5.2 shall be officers and employees of the Corporation or one of its Subsidiaries. The Administrator approving Performance-Based Awards or making any certification required pursuant to Section 5.2.4 must be constituted as provided in Section 3.1 for awards that are intended as performance-based compensation under Section 162(m) of the Code
- 5.2.2 Performance Goals. The specific performance goals for Performance-Based Awards (other than Qualifying Options and Qualifying SARs) shall be, on an absolute or relative basis, established based on such business criteria as selected by the Administrator in its sole discretion ("Business Criteria"), including, but not limited to, the following: (1) earnings per share, (2) cash flow (which means cash and cash equivalents derived from either (i) net cash flow from operations or (ii) net cash flow from operations, financing and investing activities), (3) total stockholder return, (4) price per share of Common Stock, (5) gross revenue, (6) revenue growth, (7) operating income (before or after taxes), (8) net earnings (before or after interest, taxes, depreciation and/or amortization), (9) return on equity, (10) capital employed, or on assets or on net investment, (11) cost containment or reduction, (12) cash cost per ounce of production, (13) operating margin, (14) debt reduction, (15) resource amounts, (16) production or production growth, (17) resource replacement or resource growth, (18) successful completion of financings, or (19) any combination of the foregoing. To qualify awards as performancebased under Section 162(m), the applicable Business Criterion (or Business Criteria, as the case may be) and specific performance goal or goals ("targets") must be established and approved by the Administrator during the first 90 days of the performance period (and, in the case of performance periods of less than one (1) year, in no event after 25% or more of the performance period has elapsed) and while performance relating to such target(s) remains substantially uncertain within the meaning of Section 162(m) of the Code. Performance targets shall be adjusted to mitigate the unbudgeted impact of material, unusual or nonrecurring gains and losses, accounting changes or other extraordinary events not foreseen at the time the targets were set unless the Administrator provides otherwise at the time of establishing the targets; provided that the Administrator may not make any adjustment to the extent it would adversely affect the qualification of any compensation payable under such performance targets as "performance-based compensation" under Section 162(m) of Code. The applicable performance measurement period may not be less than three (3) months nor more than ten (10) years.
- **5.2.3** Form of Payment. Grants or awards intended to qualify under this Section 5.2 may be paid in cash or shares of Common Stock or any combination thereof.
- **5.2.4** *Certification of Payment.* Before any Performance-Based Award under this Section 5.2 (other than Qualifying Options and Qualifying SARs) is paid and to the extent required to qualify the award as performance-based compensation within the meaning of Section 162(m) of the Code, the Administrator must certify in writing that the performance target(s) and any other material terms of the Performance-Based Award were in fact timely satisfied.
- **5.2.5** Reservation of Discretion. The Administrator will have the discretion to determine the restrictions or other limitations of the individual awards granted under this Section 5.2 including the authority to reduce awards, payouts or vesting or to pay no awards, in its sole discretion, if the Administrator preserves such authority at the time of grant by language to this effect in its authorizing resolutions or otherwise.

- **5.2.6** Expiration of Grant Authority. As required pursuant to Section 162(m) of the Code and the regulations promulgated thereunder, the Administrator's authority to grant new awards that are intended to qualify as performance-based compensation within the meaning of Section 162(m) of the Code (other than Qualifying Options and Qualifying SARs) shall terminate upon the first meeting of the Corporation's stockholders that occurs in the fifth (5th) year following the year in which the Corporation's stockholders first approve this Plan (the "162(m) Term").
- **5.2.7** Compensation Limitations. The maximum aggregate number of shares of Common Stock that may be issued to any Eligible Person during the term of this Plan pursuant to Qualifying Options and Qualifying SARs may not exceed 20% of the shares of Common Stock issuable under the Plan. The maximum aggregate number of shares of Common Stock that may be issued to any Eligible Person pursuant to Performance-Based Awards granted during the 162(m) Term (other than cash awards granted pursuant to Section 5.1.6 and Qualifying Options or Qualifying SARs) may not exceed 1,000,000 shares of Common Stock. The maximum amount that may be paid to any Eligible Person pursuant to Performance-Based Awards granted pursuant to Section 5.1.6 (cash awards) during the 162(m) Term may not exceed \$1,000,000.
- **5.3** Award Agreements. Each award shall be evidenced by a written or electronic award agreement in the form approved by the Administrator and, if required by the Administrator, executed by the recipient of the award. The Administrator may authorize any officer of the Corporation (other than the particular award recipient) to execute any or all award agreements on behalf of the Corporation (electronically or otherwise). The award agreement shall set forth the material terms and conditions of the award as established by the Administrator consistent with the express limitations of this Plan.
- 5.4 Deferrals and Settlements. Payment of awards may be in the form of cash, Common Stock, other awards or combinations thereof as the Administrator shall determine, and with such restrictions as it may impose. The Administrator may also require or permit participants to elect to defer the issuance of shares of Common Stock or the settlement of awards in cash under such rules and procedures as it may establish under this Plan. The Administrator may also provide that deferred settlements include the payment or crediting of interest or other earnings on the deferral amounts, or the payment or crediting of dividend equivalents where the deferred amounts are denominated in shares. All mandatory or elective deferrals of the issuance of shares of Common Stock or the settlement of cash awards shall be structured in a manner that is intended to comply with the requirements of Section 409A of the Code.
- **5.5** Consideration for Common Stock or Awards. The purchase price for any award granted under this Plan or the Common Stock to be delivered pursuant to an award, as applicable, may be paid by means of any lawful consideration as determined by the Administrator and subject to compliance with applicable laws and the rules and policies of any stock exchange (including the TSX) on which the shares of Common Stock are then listed, including, without limitation, one or a combination of the following methods:
 - a) cash, check payable to the order of the Corporation, or electronic funds transfer;
 - b) notice and third party payment in such manner as may be authorized by the Administrator;
 - c) the delivery of previously owned shares of Common Stock that are fully vested and unencumbered;
 - d) by a reduction in the number of shares otherwise deliverable pursuant to the award ('Net Exercise"); or
 - e) subject to such procedures as the Administrator may adopt, pursuant to a "cashless exercise" with a third party who provides financing for the purposes of (or who otherwise facilitates) the purchase or exercise of awards ("Cashless Exercise"),

provided that Net Exercise and Cashless Exercise alternatives for payment may only be permitted in accordance with the procedures contemplated by the policies of the TSX so long as the shares of Common Stock are listed on the TSX.

In the event that the Administrator allows a participant to exercise an award by delivering shares of Common Stock previously owned by such participant and unless otherwise expressly provided by the Administrator, any shares delivered which were initially acquired by the participant from the Corporation (upon exercise of a stock option or otherwise) must have been owned by the participant at least six (6) months as of the date of delivery (or such other period as may be required by the Administrator in order to avoid adverse accounting treatment). Shares of Common Stock used to satisfy the exercise price of an option shall be valued at their Fair Market Value on the date of exercise. The Corporation will not be obligated to deliver any shares unless and until it receives full payment of the exercise or purchase price therefor and any related withholding obligations under Section 8.5 and any other conditions to exercise or purchase, as established from time to time by the Administrator, have been satisfied. Unless otherwise expressly provided in the applicable award agreement, the Administrator may at any time eliminate or limit a participant's ability to pay the purchase or exercise price of any award by any method other than cash payment to the Corporation.

- 5.6 Definition of Fair Market Value. For purposes of this Plan "Fair Market Value" shall mean, unless otherwise determined or provided by the Administrator in the circumstances, the closing price for a share of Common Stock on the trading day immediately before the grant date, as furnished by the principal stock market or exchange on which the Common Stock is then traded for the date in question, provided that so long as the shares of Common Stock are listed on the TSX, the Fair Market Value shall not be lower than such closing price for the date in question as furnished by the TSX. If the Common Stock is no longer traded on any stock exchange or market as of the applicable date, the Fair Market Value of the Common Stock shall be the value as reasonably determined by the Administrator for purposes of the award in the circumstances.
- 5.7 Transfer Restrictions. Unless otherwise expressly provided in (or pursuant to) this Plan and permitted by applicable law and the rules and policies of any stock exchange on which the shares of Common Stock are then listed (including the TSX), (a) all awards are non-assignable and non-transferable and shall not be subject in any manner to sale, transfer, anticipation, alienation, assignment, pledge, encumbrance or charge; (b) awards shall be exercised only by the participant; and (c) amounts payable or shares issuable pursuant to any award shall be delivered only to (or for the account of) the participant.
- **5.8** *International Awards*. One or more awards may be granted to Eligible Persons who provide services to the Corporation or one of its Subsidiaries outside of the United States. Any awards granted to such persons may, if deemed necessary or advisable by the Administrator, be granted pursuant to the terms and conditions of any applicable sub-plans, if any, appended to this Plan and approved by the Administrator.

5.9 Vesting.

- **5.9.1** Subject to Sections 5.1.2 and 5.9.2 hereof, awards shall vest at such time or times and subject to such terms and conditions as shall be determined by the Administrator at the time of grant; provided, however, that in the absence of any award vesting periods designated by the Administrator at the time of grant in the applicable award agreement, awards shall vest as to one-fourth of the total number of shares subject to the award on each of the first, second, third and fourth anniversaries of the date of grant.
- 5.9.2 No award granted under this Plan, other than options, may vest before the date that is one year following the date of grant or issue. In accordance with TSX policies, and subject to TSX approval to the contrary, and provided that the shares of Common Stock are then listed on the TSX, options granted to investor relations service providers must vest (and not otherwise be exercisable) in stages over a minimum of twelve (12) months, such that:
 - (a) no more than ¼ of the options vest no sooner than three (3) months after the options were granted;
 - (b) no more than another ½ of the options vest no sooner than six (6) months after the options were granted;
 - (c) no more than another 1/4 of the options vest no sooner than nine (9) months after the options were granted; and
 - (d) the remainder of the options vest no sooner than twelve (12) months after the options were granted.

6. EFFECT OF TERMINATION OF SERVICE ON AWARDS

6.1 Termination of Employment.

- **6.1.1** The Administrator shall establish the effect of a termination of employment or service on the rights and benefits under each award under this Plan and in so doing may make distinctions based upon, inter alia, the cause of termination and type of award. If the participant is not an employee of the Corporation or one of its Subsidiaries and provides other services to the Corporation or one of its Subsidiaries, the Administrator shall be the sole judge for purposes of this Plan (unless a contract or the award agreement otherwise provides) of whether the participant continues to render services to the Corporation or one of its Subsidiaries and the date, if any, upon which such services shall be deemed to have terminated.
- 6.1.2 Subject to Section 6.4, for awards of stock options or SARs, unless the award agreement provides otherwise, the exercise period of such options or SARs shall expire: (1) three (3) months after the last day that the participant is employed by or provides services to the Corporation or a Subsidiary (provided; however, that in the event of the participant's death during this period, those persons entitled to exercise the option or SAR pursuant to the laws of descent and distribution shall have one (1) year following the date of death within which to exercise such option or SAR); (2) in the case of a participant whose termination of employment is due to death or disability (as defined in the applicable award agreement), twelve (12) months after the last day that the participant is employed by or provides services to the Corporation or a Subsidiary; and (3) immediately upon a participant's termination for "cause". The Administrator will, in its absolute discretion, determine the effect of all matters and questions relating to a termination of employment, including, but not by way of limitation, the question of whether a leave of absence constitutes a termination of employment and whether a participant's termination is for "cause."

If not defined in the applicable award agreement, "cause" shall mean:

- (i) conviction of a felony or a crime involving fraud or moral turpitude; or
- (ii) theft, material act of dishonesty or fraud, intentional falsification of any employment or Corporation records, or commission of any criminal act which impairs participant's ability to perform appropriate employment duties for the Corporation; or
- (iii) intentional or reckless conduct or gross negligence materially harmful to the Corporation or the successor to the Corporation after a Change in Control, including violation of a non-competition or confidentiality agreement; or
 - (iv) willful failure to follow lawful instructions of the person or body to which participant reports; or
 - (v) gross negligence or willful misconduct in the performance of participant's assigned duties.

Cause shall not include mere unsatisfactory performance in the achievement of participant's job objectives.

- 6.1.3 For awards of RSUs, unless the award agreement provides otherwise, RSUs that are subject to restrictions at the time that a participant whose employment or service is terminated shall be forfeited to the Corporation; provided that, the Administrator may provide, by rule or regulation or in any award agreement, or may determine in any individual case, that restrictions or forfeiture conditions relating to RSUs shall be waived in whole or in part in the event of terminations resulting from specified causes, and the Administrator may in other cases waive in whole or in part the forfeiture of RSUs.
- **6.2** Events Not Deemed Terminations of Service. Unless the express policy of the Corporation or one of its Subsidiaries, or the Administrator, otherwise provides, the employment relationship shall not be considered terminated in the case of (a) sick leave, (b) military leave or (c) any other leave of absence authorized by the Corporation or one of its Subsidiaries, or the Administrator; provided that unless reemployment upon the expiration of such leave is guaranteed by contract or law, such leave is for a period of not more than three (3) months. In the case of any employee of the Corporation or one of its Subsidiaries on an approved leave of absence, continued vesting of the award while on leave from the employ of the Corporation or one of its Subsidiaries may be suspended until the employee returns to service, unless the Administrator otherwise provides or applicable law otherwise requires. In no event shall an award be exercised after the expiration of the term set forth in the award agreement.
- **6.3** Effect of Change of Subsidiary Status. For purposes of this Plan and any award, if an entity ceases to be a Subsidiary of the Corporation, a termination of employment or service shall be deemed to have occurred with respect to each Eligible Person in respect of such Subsidiary who does not continue as an Eligible Person in respect of another entity within the Corporation or another Subsidiary that continues as such after giving effect to the transaction or other event giving rise to the change in status.

6.4 Expiry. Notwithstanding the foregoing, any award granted or issued to a participant who is a director, officer, employee, consultant or management company employee must expire no later than twelve (12) months following the date the participant ceases to be an Eligible Person under this Plan.

7. ADJUSTMENTS; ACCELERATION

- 7.1 Adjustments. Upon or in contemplation of any of the following events described in this Section 7.1: any reclassification, recapitalization, stock split (including a stock split in the form of a stock dividend) or reverse stock split ("stock split"); any merger, arrangement, combination, consolidation or other reorganization; any spin-off, split-up or similar extraordinary dividend distribution in respect of the Common Stock (whether in the form of securities or property); any exchange of Common Stock or other securities of the Corporation, or any similar, unusual or extraordinary corporate transaction in respect of the Common Stock; then the Administrator shall in such manner, to such extent and at such time as it deems appropriate and equitable in the circumstances (but subject to compliance with applicable laws and stock exchange requirements) proportionately adjust any or all of (1) the number and type of shares of Common Stock (or other securities) that thereafter may be made the subject of awards (including the number of shares provided for in this Plan), (2) the number, amount and type of shares of Common Stock (or other securities or property) subject to any or all outstanding awards, (3) the grant, purchase, or exercise price (which term includes the base price of any SAR or similar right) of any or all outstanding awards, (4) the securities, cash or other property deliverable upon exercise or payment of any outstanding awards, and (5) the 162(m) compensation limitations set forth in Section 5.2.7 and (subject to Section 8.8.3(a)) the performance standards applicable to any outstanding awards (provided that no adjustment shall be allowed to the extent inconsistent with the requirements of Code section 162(m)). Any adjustment made pursuant to this Section 7.1 shall be made in a manner that, in the good faith determination of the Administrator, will not likely result in the imposition of additional taxes or interest under Section 409A of the Code. With respect to any award of an ISO, the Administrator
- 7.2 Change in Control. Upon a Change in Control, each then-outstanding option and SAR shall automatically become fully vested and each other award granted under this Plan that is then outstanding shall automatically become vested and payable to the holder of such award unless the Administrator has made appropriate provision for the substitution, assumption, exchange or other continuation of the award pursuant to the Change in Control. Notwithstanding the foregoing, the Administrator, in its sole and absolute discretion, may choose (in an award agreement or otherwise) to provide for full or partial accelerated vesting of any award upon a Change In Control (or upon any other event or other circumstance related to the Change in Control, such as an involuntary termination of employment occurring after such Change in Control, as the Administrator may determine), irrespective of whether such any such award has been substituted, assumed, exchanged or otherwise continued pursuant to the Change in Control.

For purposes of this Plan, "Change in Control" shall be deemed to have occurred if:

(i) a tender offer (or series of related offers) shall be made and consummated for the ownership of 50% or more of the outstanding voting securities of the Corporation, unless as a result of such tender offer more than 50% of the outstanding voting securities of the surviving or resulting corporation shall be owned in the aggregate by the stockholders of the Corporation (as of the time immediately prior to the commencement of such offer), any employee benefit plan of the Corporation or its Subsidiaries, and their affiliates;

- (ii) the Corporation shall be merged or consolidated with another entity, unless as a result of such merger or consolidation more than 50% of the outstanding voting securities of the surviving or resulting entity shall be owned in the aggregate by the stockholders of the Corporation (as of the time immediately prior to such transaction), any employee benefit plan of the Corporation or its Subsidiaries, and their affiliates;
- (iii) the Corporation shall sell substantially all of its assets to another entity that is not wholly owned by the Corporation, unless as a result of such sale more than 50% of such assets shall be owned in the aggregate by the stockholders of the Corporation (as of the time immediately prior to such transaction), any employee benefit plan of the Corporation or its Subsidiaries and their affiliates; or
- (iv) a Person (as defined below) shall acquire 50% or more of the outstanding voting securities of the Corporation (whether directly, indirectly, beneficially or of record), unless as a result of such acquisition more than 50% of the outstanding voting securities of the surviving or resulting corporation shall be owned in the aggregate by the stockholders of the Corporation (as of the time immediately prior to the first acquisition of such securities by such Person), any employee benefit plan of the Corporation or its Subsidiaries, and their affiliates.

For purposes of this Section 7.2, ownership of voting securities shall take into account and shall include ownership as determined by applying the provisions of Rule 13d-3(d)(I)(i) (as in effect on the date hereof) under the *Exchange Act*. In addition, for such purposes, "Person" shall have the meaning given in Section 3(a)(9) of the *Exchange Act*, as modified and used in Sections 13(d) and 14(d) thereof; provided, however, that a Person shall not include (A) the Corporation or any of its Subsidiaries; (B) a trustee or other fiduciary holding securities under an employee benefit plan of the Corporation or any of its Subsidiaries; (C) an underwriter temporarily holding securities pursuant to an offering of such securities; or (D) a corporation owned, directly or indirectly, by the stockholders of the Corporation in substantially the same proportion as their ownership of stock of the Corporation.

7.3 Early Termination of Awards. Any award that has been accelerated as required or permitted by Section 7.2 upon a Change in Control (or would have been so accelerated but for Section 7.4 or 7.5) shall terminate upon such event, subject to any provision that has been expressly made by the Administrator, through a plan of reorganization or otherwise, for the survival, substitution, assumption, exchange or other continuation of such award and provided that, in the case of options and SARs that will not survive, be substituted for, assumed, exchanged, or otherwise continued in the transaction, the holder of such award shall be given reasonable advance notice of the impending termination and a reasonable opportunity to exercise his or her outstanding options and SARs in accordance with their terms before the termination of such awards (except that in no case shall more than ten (10) days' notice of accelerated vesting and the impending termination be required and any acceleration may be made contingent upon the actual occurrence of the event).

The Administrator may make provision for payment in cash or property (or both) in respect of awards terminated pursuant to this section as a result of the Change in Control and may adopt such valuation methodologies for outstanding awards as it deems reasonable and, in the case of options, SARs or similar rights, and without limiting other methodologies, may base such settlement solely upon the excess if any of the per share amount payable upon or in respect of such event over the exercise or base price of the award.

- **7.4 Other Acceleration Rules.** Any acceleration of awards pursuant to this Section 7 shall comply with applicable legal and stock exchange requirements and, if necessary to accomplish the purposes of the acceleration or if the circumstances require, may be deemed by the Administrator to occur a limited period of time not greater than thirty (30) days before the event. Without limiting the generality of the foregoing, the Administrator may deem an acceleration to occur immediately prior to the applicable event and/or reinstate the original terms of an award if an event giving rise to the acceleration does not occur. Notwithstanding any other provision of the Plan to the contrary, the Administrator may override the provisions of Section 7.2, 7.3, and/or 7.5 by express provision in the award agreement or otherwise. The portion of any ISO accelerated pursuant to Section 7.2 or any other action permitted hereunder shall remain exercisable as an ISO only to the extent the applicable \$100,000 limitation on ISOs is not exceeded. To the extent exceeded, the accelerated portion of the option shall be exercisable as a nonqualified stock option under the Code.
- **7.5** *Possible Rescission of Acceleration*. If the vesting of an award has been accelerated expressly in anticipation of an event and the Administrator later determines that the event will not occur, the Administrator may rescind the effect of the acceleration as to any then outstanding and unexercised or otherwise unvested awards; *provided, that*, in the case of any compensation that has been deferred for purposes of Section 409A of the Code, the Administrator determines that such rescission will not likely result in the imposition of additional tax or interest under Code Section 409A.

8. OTHER PROVISIONS

- 8.1 Compliance with Laws. This Plan, the granting and vesting of awards under this Plan, the offer, issuance and delivery of shares of Common Stock, the acceptance of promissory notes and/or the payment of money under this Plan or under awards are subject to compliance with all applicable federal and state laws, rules and regulations (including but not limited to state and federal securities law, federal margin requirements), and rules and policies of any stock exchange upon which the Common Stock are listed (including the TSX), and to such approvals by any applicable stock exchange listing, regulatory or governmental authority as may, in the opinion of counsel for the Corporation, be necessary or advisable in connection therewith. The person acquiring any securities under this Plan will, if requested by the Corporation or one of its Subsidiaries, provide such assurances and representations to the Corporation or one of its Subsidiaries as the Administrator may deem necessary or desirable to assure compliance with all applicable legal and accounting requirements.
- **8.2** Future Awards/Other Rights. No person shall have any claim or rights to be granted an award (or additional awards, as the case may be) under this Plan, subject to any express contractual rights (set forth in a document other than this Plan) to the contrary.
- **8.3** No Employment/Service Contract. Nothing contained in this Plan (or in any other documents under this Plan or in any award) shall confer upon any Eligible Person or other participant any right to continue in the employ or other service of the Corporation or one of its Subsidiaries, constitute any contract or agreement of employment or other service or affect an employee's status as an employee at will or shall interfere in any way with the right of the Corporation or one of its Subsidiaries to change a person's compensation or other benefits, or to terminate his or her employment or other service, with or without cause. Nothing in this Section 8.3, however, is intended to adversely affect any express independent right of such person under a separate employment or service contract other than an award agreement.

- **8.4 Plan Not Funded.** Awards payable under this Plan shall be payable in shares or from the general assets of the Corporation, and no special or separate reserve, fund or deposit shall be made to assure payment of such awards. No participant, beneficiary or other person shall have any right, title or interest in any fund or in any specific asset (including shares of Common Stock, except as expressly otherwise provided) of the Corporation or one of its Subsidiaries by reason of any award hereunder. Neither the provisions of this Plan (or of any related documents), nor the creation or adoption of this Plan, nor any action taken pursuant to the provisions of this Plan shall create, or be construed to create, a trust of any kind or a fiduciary relationship between the Corporation or one of its Subsidiaries and any participant, beneficiary or other person. To the extent that a participant, beneficiary or other person acquires a right to receive payment pursuant to any award hereunder, such right shall be no greater than the right of any unsecured general creditor of the Corporation.
- **8.5** Tax Withholding. Upon any exercise, vesting, or payment of any award, the Corporation or one of its Subsidiaries shall have the right at its option to:
- (a) require the participant (or the participant's personal representative or beneficiary, as the case may be) to pay or provide for payment of at least the minimum amount of any taxes which the Corporation or one of its Subsidiaries may be required to withhold with respect to such award event or payment; or
- (b) deduct from any amount otherwise payable in cash to the participant (or the participant's personal representative or beneficiary, as the case may be) the minimum amount of any taxes which the Corporation or one of its Subsidiaries may be required to withhold with respect to such cash payment.

In any case where a tax is required to be withheld in connection with the delivery of shares of Common Stock under this Plan, the Administrator may in its sole discretion (subject to Section 8.1) grant (either at the time of the award or thereafter) to the participant the right to elect, pursuant to such rules and subject to such conditions as the Administrator may establish, to have the Corporation reduce the number of shares to be delivered by (or otherwise reacquire) the appropriate number of shares, valued in a consistent manner at their Fair Market Value or at the sales price in accordance with authorized procedures for cashless exercises, necessary to satisfy the minimum applicable withholding obligation on exercise, vesting or payment. In no event shall the shares withheld exceed the minimum whole number of shares required for tax withholding under applicable law.

8.6 Effective Date, Termination and Suspension, Amendments.

- **8.6.1** Effective Date and Termination. This Plan was approved by the Board and became effective on February 9, 2017. Unless earlier terminated by the Board, this Plan shall terminate at the close of business on February 9, 2027. After the termination of this Plan either upon such stated expiration date or its earlier termination by the Board, no additional awards may be granted under this Plan, but previously granted awards (and the authority of the Administrator with respect thereto, including the authority to amend such awards) shall remain outstanding in accordance with their applicable terms and conditions and the terms and conditions of this Plan.
- **8.6.2** *Authorization.* Subject to obtaining the necessary regulatory approvals or stockholder approval as required pursuant to the rules and policies of the TSX, as in effect from time to time, so long as the shares of Common Stock are listed on the TSX, the Board may, at any time, terminate or, from time to time, amend, modify or suspend this Plan, in whole or in part. No awards may be granted during any period that the Board suspends this Plan.

- **8.6.3** *Stockholder Approval.* To the extent then required by applicable law or any applicable stock exchange or required under Sections 162, 422 or 424 of the Code to preserve the intended tax consequences of this Plan, or deemed necessary or advisable by the Board, this Plan and any amendment to this Plan shall be subject to stockholder approval.
- **8.6.4** *Amendments to Awards.* Without limiting any other express authority of the Administrator under (but subject to) the express limits of this Plan, the Administrator by agreement or resolution may waive conditions of or limitations on awards to participants that the Administrator in the prior exercise of its discretion has imposed, without the consent of a participant, and (subject to the requirements of Sections 3.2 and 8.6.5) may make other changes to the terms and conditions of awards. Any amendment or other action that would constitute a repricing of an award is subject to the limitations set forth in Section 3.2(g).

8.6.5 Limitations on Amendments to Plan and Awards

- (a) No amendment, suspension or termination of this Plan or change of or affecting any outstanding award shall, without written consent of the participant, affect in any manner materially adverse to the participant any rights or benefits of the participant or obligations of the Corporation under any award granted under this Plan prior to the effective date of such change. Changes, settlements and other actions contemplated by Section 7 shall not be deemed to constitute changes or amendments for purposes of this Section 8.6.
- (b) Notwithstanding any other provision of this Plan, disinterested stockholder approval must be obtained for any reduction in the exercise price of a stock option, or the extension of the term of a stock option, if the participant is an insider of the Corporation at the time of the proposed amendment.
- **8.7** *Privileges of Stock Ownership.* Except as otherwise expressly authorized by the Administrator or this Plan, a participant shall not be entitled to any privilege of stock ownership as to any shares of Common Stock not actually delivered to and held of record by the participant. No adjustment will be made for dividends or other rights as a stockholder for which a record date is prior to such date of delivery.

8.8 Governing Law; Construction; Severability.

- **8.8.1** Choice of Law. This Plan, the awards, all documents evidencing awards and all other related documents shall be governed by, and construed in accordance with the laws of the State of Delaware.
- **8.8.2** Severability. If a court of competent jurisdiction holds any provision invalid and unenforceable, the remaining provisions of this Plan shall continue in effect.

8.8.3 Plan Construction.

(a) Rule 16b-3. It is the intent of the Corporation that the awards and transactions permitted by awards be interpreted in a manner that, in the case of participants who are or may be subject to Section 16 of the Exchange Act, qualify, to the maximum extent compatible with the express terms of the award, for exemption from matching liability under Rule 16b-3 promulgated under the Exchange Act. Notwithstanding the foregoing, the Corporation shall have no liability to any participant for Section 16 consequences of awards or events under awards if an award or event does not so qualify.

- (b) Section 162(m). Awards under Sections 5.1.5 through 5.1.6 to persons described in Section 5.2 that are either granted or become vested, exercisable or payable based on attainment of one or more performance goals related to the Business Criteria, as well as Qualifying Options and Qualifying SARs granted to persons described in Section 5.2, that are approved by a committee composed solely of two or more outside directors (as this requirement is applied under Section 162(m) of the Code) shall be deemed to be intended as performance-based compensation within the meaning of Section 162(m) of the Code unless such committee provides otherwise at the time of grant of the award. It is the further intent of the Corporation that (to the extent the Corporation or one of its Subsidiaries or awards under this Plan may be or become subject to limitations on deductibility under Section 162(m) of the Code) any such awards and any other Performance-Based Awards under Section 5.2 that are granted to or held by a person subject to Section 162(m) will qualify as performance-based compensation or otherwise be exempt from deductibility limitations under Section 162(m).
- (c) Code Section 409A Compliance. The Board intends that, except as may be otherwise determined by the Administrator, any awards under the Plan are either exempt from or satisfy the requirements of Section 409A of the Code and related regulations and Treasury pronouncements ("Section 409A") to avoid the imposition of any taxes, including additional income or penalty taxes, thereunder. If the Administrator determines that an award, award agreement, acceleration, adjustment to the terms of an award, payment, distribution, deferral election, transaction or any other action or arrangement contemplated by the provisions of the Plan would, if undertaken, cause a participant's award to become subject to Section 409A, unless the Administrator expressly determines otherwise, such award, award agreement, payment, acceleration, adjustment, distribution, deferral election, transaction or other action or arrangement shall not be undertaken and the related provisions of the Plan and/or award agreement will be deemed modified or, if necessary, rescinded in order to comply with the requirements of Section 409A to the extent determined by the Administrator without the content or notice to the participant. Notwithstanding the foregoing, neither the Corporation nor the Administrator shall have any obligation to take any action to prevent the assessment of any excise tax or penalty on any participant under Section 409A and neither the Corporation nor the Administrator will have any liability to any participant for such tax or penalty.
- (d) No Guarantee of Favorable Tax Treatment Although the Corporation intends that awards under the Plan will be exempt from, or will comply with, the requirements of Section 409A of the Code, the Corporation does not warrant that any award under the Plan will qualify for favorable tax treatment under Section 409A of the Code or any other provision of federal, state, local or foreign law. The Corporation shall not be liable to any participant for any tax, interest or penalties the participant might owe as a result of the grant, holding, vesting, exercise or payment of any award under the Plan
- **8.9** Captions. Captions and headings are given to the sections and subsections of this Plan solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of this Plan or any provision thereof.

- 8.10 Stock-Based Awards in Substitution for Stock Options or Awards Granted by Other Corporation. Awards may be granted to Eligible Persons in substitution for or in connection with an assumption of employee stock options, SARs, or other stock-based awards granted by other entities to persons who are or who will become Eligible Persons in respect of the Corporation or one of its Subsidiaries, in connection with a distribution, arrangement, business combination, merger or other reorganization by or with the granting entity or an affiliated entity, or the acquisition by the Corporation or one of its Subsidiaries, directly or indirectly, of all or a substantial part of the stock or assets of the employing entity. The awards so granted need not comply with other specific terms of this Plan, provided the awards reflect only adjustments giving effect to the assumption or substitution consistent with the conversion applicable to the Common Stock in the transaction and any change in the issuer of the security. Any shares that are delivered and any awards that are granted by, or become obligations of, the Corporation, as a result of the assumption by the Corporation of, or in substitution for, outstanding awards previously granted by an acquired company (or previously granted by a predecessor employer (or direct or indirect parent thereof) in the case of persons that become employed by the Corporation or one of its Subsidiaries in connection with a business or asset acquisition or similar transaction) shall not be counted against the Share Limit or other limits on the number of shares available for issuance under this Plan, except as may otherwise be provided by the Administrator at the time of such assumption or substitution or as may be required to comply with the requirements of any applicable stock exchange.
- **8.11** Non-Exclusivity of Plan. Nothing in this Plan shall limit or be deemed to limit the authority of the Board or the Administrator to grant awards or authorize any other compensation, with or without reference to the Common Stock, under any other plan or authority.
- 8.12 No Corporate Action Restriction. The existence of this Plan, the award agreements and the awards granted hereunder shall not limit, affect or restrict in any way the right or power of the Board or the stockholders of the Corporation to make or authorize: (a) any adjustment, recapitalization, reorganization or other change in the capital structure or business of the Corporation or any Subsidiary, (b) any merger, arrangement, business combination, amalgamation, consolidation or change in the ownership of the Corporation or any Subsidiary, (c) any issue of bonds, debentures, capital, preferred or prior preference stock ahead of or affecting the capital stock (or the rights thereof) of the Corporation or any Subsidiary, (d) any dissolution or liquidation of the Corporation or any Subsidiary, (e) any sale or transfer of all or any part of the assets or business of the Corporation or any Subsidiary, or (f) any other corporate act or proceeding by the Corporation or any Subsidiary. No participant, beneficiary or any other person shall have any claim under any award or award agreement against any member of the Board or the Administrator, or the Corporation or any employees, officers or agents of the Corporation or any Subsidiary, as a result of any such action.
- **8.13** Other Corporation Benefit and Compensation Programs. Payments and other benefits received by a participant under an award made pursuant to this Plan shall not be deemed a part of a participant's compensation for purposes of the determination of benefits under any other employee welfare or benefit plans or arrangements, if any, provided by the Corporation or any Subsidiary, except where the Administrator expressly otherwise provides or authorizes in writing or except as otherwise specifically set forth in the terms and conditions of such other employee welfare or benefit plan or arrangement. Awards under this Plan may be made in addition to, in combination with, as alternatives to or in payment of grants, awards or commitments under any other plans or arrangements of the Corporation or its Subsidiaries.
- **8.14** *Prohibition on Repricing.* Subject to Section 4, the Administrator shall not, without the approval of the stockholders of the Corporation (i) reduce the exercise price, or cancel and reissue options so as to in effect reduce the exercise price or (ii) change the manner of determining the exercise price so that the exercise price is less than the Fair Market Value per share of Common Stock.



MAJORITY VOTING POLICY

The Toronto Stock Exchange ("TSX") has certain rules and regulations around the election of directors that necessitate the adoption of a Majority Voting Policy.

Specifically, the TSX requires:

- TSX listed issuers ("Listed Issuers") elect Directors individually;
- Listed Issuers hold annual elections for all Directors;
- each Director of a Listed Issuer must be elected by a majority (50% +1) of the votes cast with respect to his/her election ("TSX Majority Voting Requirement"), other than at contested meetings;
- Listed Issuers must adopt a majority voting policy containing the requirements prescribed under section 461.3 of the TSX Manual, unless such Listed Issuer otherwise satisfies the Majority Voting Requirement in a manner acceptable to the TSX;
- Listed Issuers must fully describe their majority voting policy on an annual basis in the materials which they send to their shareholders in connection with a meeting at which Directors are being elected; and
- following each meeting of security holders at which there is a vote on the election of Directors at an uncontested meeting, each Listed Issuer must forthwith issue a news release disclosing the detailed voting results for the election of each Director, which news release must disclose one of the following: (i) the percentages of votes received "for" and "withheld" for each Director; (ii) the total votes cast by ballot with the number that each Director received "for"; or "(iii) the percentages and total number of votes received "for" each Director.

The Board of Directors (the "Board") of Birchtech Corp. (the "Company") has unanimously adopted this majority voting policy (the "Policy") and each of the current Directors of the Company as well as future nominees for election to the Board will be required to confirm that they will abide by this Policy.

Pursuant to this Policy, the forms of proxy for the election of Directors will permit the shareholders of the Company (each a "Shareholder" and collectively, the "Shareholders") to vote in favor of, or to withhold from voting, separately, for each Director nominee. The Chairperson of the Board will ensure that the number of votes in favor of or withheld from voting for each Director nominee is recorded and promptly made public in accordance with TSX guidelines after each Shareholder meeting at which Directors are elected. Neither broker non-votes nor abstentions shall be deemed to be voted "withheld" from a Director's election. If a nominee Director is not elected in accordance with the TSX Majority Voting Requirement at a Shareholder meeting, then, for the purposes of this Policy, the Director nominee shall be considered not to have received the support of the Shareholders, even though duly elected as a matter of corporate law.

Immediately following the certification of the Shareholder vote of a Shareholder meeting at which a Director nominee elected as a Director is considered under this test not to have the confidence of the Shareholders, the Director nominee must forthwith submit his/her resignation (the "Resignation") to the Board. The Resignation will not be effective until accepted by the Board.

The Board shall consider all factors deemed relevant by members of the Board including, without limitation, the underlying reasons (if ascertainable), why Shareholders withheld votes from the election of that Director nominee, the length of service and the qualifications of such Director's contributions to the Company and the Company's governance guidelines. Within ninety (90) days following the applicable Shareholders' meeting, the Board shall conclude its deliberations and shall make a determination as to whether or not to accept the Resignation, however, as mandated in the TSX Manual, the Board shall accept the Resignation absent exceptional circumstances.

Following the Board's determination in this regard, the Board shall publicly disclose their decision to accept or not to accept the Resignation, including, if applicable, the reasons for not accepting the Resignation of such Director.

If a Resignation is tendered and accepted pursuant to this Policy, subject to any applicable corporate law restrictions, the Board may leave the vacancy unfilled or may appoint a new Director to fill the vacancy. A Director who tenders a Resignation pursuant to this Policy shall not be permitted to participate in any meetings of the Board or any sub-committee of the Board at which his/her election as a Director is being considered.

In the event that a majority of the Company's Directors are not elected in accordance with the TSX Majority Voting Requirement at the same Shareholder meeting, then such Directors not receiving such required majority vote shall not be permitted to vote in any meeting of the Board at which his/ her election is being considered, however he/she shall be counted for the purpose of determining whether the Board has a quorum at such meeting.

The Board may adopt such procedures as they see fit to assist it in their determinations with respect to this Policy. This Policy does not apply where an election involves a proxy battle, i.e., where proxy material is circulated in support of one or more nominees who are not part of the Director nominees supported by the Board.

Please direct your questions as to any of the matters discussed in this Policy to the Company's Chief Legal Officer or General Counsel.

Adopted by the Board of Directors of Birchtech Corp. effective October 30, 2024.

